

AMOR IX, LLC Soda Lake Repower Property/Sales & Use Tax Abatements Application October 2018

Compiled By: Cory Draper 136 S Main St STE 600 Salt Lake City, UT 84101 (O) – 801.875.4200 cory.draper@cyrqenergy.com



Table of Contents

- 1. Facility Information
 - a. Description of Technology and Facility
 - b. Legal Descriptions & Maps
 - c. Business Plan
 - d. Tax Records
 - e. Business License
- 2. Facility Permits
- 3. Contractors & Subcontractors
- 4. Employment Information
- 5. Supplemental Information
- 6. Tax Schedules
- 7. Letter regarding Power Purchase Agreement
- 8. Signature & Attestation
- 9. Confidential Information

State of Nevada Renewable Energy Tax Abatement Application

	AFN:
СН	ECKLIST - PLEASE ATTACH:
	Description of the Technology and Complete Facility including generation, transmission or distribution, the physical point at which the ownership of energy is transferred and nature of the connection to the transmission grid
2	Complete and legal description of the location of the proposed facility, including a regional facility map that identifies the location, county boundaries and state boundaries of the proposed facility or a reference to any such map of appropriate scale
3	Description of any natural or nonrenewable resources that will be affected by or required to be used in the construction or operation of the proposed facility, N/A including statement of any areas of mitigation, controversy, issue or concern
4	Summary of the PUC and FERC Dockets if any PUC and FERC filing have started
5	Copy of the Business Plan for the Nevada Facility
6	For Expansion Applications, Copy of the most recent assessment schedule and tax bill from the County Assessor's Office or the Department of Taxation
7	Website link to company profile
8	Copy of the Current Nevada State Business License
	Facility Information Form
	Employment Information, construction, and permanent employee salary schedules
	Supplemental Information Form
	Taxation Reporting Forms (Summary Sheet and Schedules 1 through 8)
	Names and contact information for construction company, contractors, subcontractors
	Letter from the utility or company describing the highlights of PPA, LOI, or MOU.
15	Confidential Information Identification Form



Attachment 1 Description of Technology & Facility

The repowered facility at Soda Lake will consist of 1 Ormat Energy Converter (OEC) unit. The inlet geothermal brine flows through the level I vaporizer where heat energy is extracted and used to convert the motive fluid to high pressure vapors which drive the level I turbine. This reduced temperature brine enters the level II vaporizer where it converts the motive fluid to vapors at lower pressure which drive the level II turbine. Both the level I and II turbines are directly coupled to the ends of the Synchronous Generator. The balance of the process consists of air-cooled condensers which condense the spent motive fluid vapors, and the preheaters and recuperator which optimize the thermodynamic process.



Attachment 2 Legal Descriptions & Maps

2-1 Legal Description and Map of Proposed Facility

Legal Description: S 1/2 SE1/4, NE 1/4 SE 1/4, E 1/2 NW 1/4 SE 1/4 SEC. 29, T. 20N, R.28E; N 1/2 SEC. 28, T. 20N, R.28E; S 1/2 SEC. 28, T.20N, R.28E; SEC. 31, T. 20N, R.28E;



0 1000 2000 3000 meters

2-2 Regional Map with Proposed Facility Location









Attachment 4 PUC & FERC Dockets Summary

AMOR IX, LLC intends to file for Market Based Rate Authority with FERC in March 2019. No PUC Filings will be necessary



Attachment 5 Business Plan



BUSINESS PLAN

Upcoming Construction

AMOR IX, LLC (Soda Lake) is a geothermal facility consisting primarily of geothermal resource, geothermal power plant, transmission line and ancillary equipment. The geothermal resource has demonstrated stable production over the past approximate 30 years and is anticipated to do so well into the future. The existing geothermal power plant was installed in 1988 and expanded in 1991. The efficiency of geothermal power plants has improved since the installment of the existing Soda Lake geothermal power plant. Soda Lake has decided to Repower the Soda Lake facility with current day, higher efficiency geothermal power plant which will increase generation from the existing geothermal resources. Without the Repower, output would be significantly lower and the continued operation of the facility over the long term may not be economically viable. Towards that end, Soda Lake has executed an EPC Contract with Ormat Nevada, Inc (Ormat) for the Soda Lake Geothermal Repower Project. A Full Notice to Proceed was issued to Ormat on July 27, 2018. The scheduled Commercial Operations Date is January 1, 2020.

Power Purchase Agreement

Soda Lake has executed a Renewable Energy Contract (Schedule 32) with PacifiCorp (dba Rocky Mountain Power) for all the generation from the Soda Lake Repower. Soda Lake will deliver generation from the Repowered Soda Lake Geothermal Power Plant to PacifiCorp who will ultimately deliver it to the University of Utah. The term of the Renewable Energy Contract is well in excess of the 10-year criterion as required by the Renewable Energy Property Tax Abatement.

Financial Performance

Soda Lake expects to generate revenues in excess of \$100 Million and sufficient cashflow to pay all expenses, keep the plant operating for the term of the Renewable Energy Contract and support an acceptable return to its investors.



Attachment 6 Tax Records



ersona ro ert a es Data ecured Tax Inquir Recorder earch						
Parcel De ail for Par	cel # 009-391-17					
Loca ion	Ownership					
ro ert Location 28-20-28 Town Add' Addresses District 2.0 - Count Tax Rate Assessor Ma s ubdivision Lot B ock Lega Descri tion ro ert Name BLM Remarks	Assessed Owner Name BUREAU OF LAND MANAGEMENT Mai ing Address 5665 MORGAN MILL RD CAR ON CITY NV 89701-1448 Lega Owner Name BUREAU OF LAND MANAGEMENT Vesting Doc #, Date Ma Document #s					
Descrip ion	Appraisal Classifica ions					
Tota Acres 640.000 quare Feet 0 Ag Acres .000 W/R Acres .000 Improvemen s	Current Land Use Code 190 Code Tab e					
ing e- 0 Non-dwe ing Units 0 Bedrooms / Baths 0 / .00	Zoning Code(s) RR20					
ing e- famiMobi e Home Hooku s 0tories .0Mu ti famiMobi e Home Hooku s 0tories .0Mu ti famiWe s 0Garage quare Ft 0Mobi e Homes 0Septic Tanks 0Attached / DetachedTota Dwe ing Units 0Buildings Sq Ft 0 Residence Sq Ft 0Residence Sq Ft 0	.0 Re-a raisa Grou 3 Re-a raisa Year 2017					
Improvement List Basement Sq Ft 0 Basement Finished Basement SF 0 Bedrooms / Baths 0/.00						
Assessed Valuation	Taxable Valuation					
Assessed Values 2018-19 2017-18 2016-17 Land 6,650 6,650 6,650 Improvements 0 0 0 Personal Property 0 0 0 Ag Land 0 0 0 Exemptions 6,650 6,650 6,650 Net Assessed Value 0 0 0	Taxable Values 2018-19 2017-18 2016-17 Land 19,000 19,000 19,000 Improvements 0 0 0 Personal Property 0 0 0 Ag Land 0 0 0 Exemptions 19,000 19,000 19,000 Net Taxable Value 0 0 0					
Increased (New) Values Land 0 0 0 Improvements 0 0 0 Personal Property 0 0 0	Increased (New) Values Land 0 0 0 Improvements 0 0 0 Personal Property 0 0 0					

Back to Search List



	erson	a roei	rt a	es Data ecur	ed Tax Inquir	Recorder earch				
			Parcel	De ail for Par	cel # 009-3	391-14				
Lo	ca ion					0	wnership			
ro ert Location 29-20-28					Assessed (Owner Name OWEL	ON ROBER	RT ET AL		
TownAdd' AddressesDistrict 2.0 - CountTax RateAssessor Ma subdivisionLotB ockLegaLegaDescri tion						i ing Address % MARIL			Ownershi I	Histor
					/	Add' Owners 5938 N W	/ALNUT GR BRIEL CA 9'		Description	
					Lena (Owner Name OWEL				
ro ert Name					-		Book / ag			
						Document #s	Doon, ag	•		
Remarks										
Dese	crip ion					Appraisa	I Classific	a ions		
Tota Acres 140.000	quare F						_			
Ag Acres .000		res .000				Current Land Use Co	ode 100 🛛	Code Tab e		
	ovemen s									
tami Detached	lwe ing U	nits <mark>0</mark>	Bedro	oms / Baths <mark>0 /</mark> .00	Zoning Code(s) A10					
ing e- 0 Mobi e Ho fami Attached	ome Hooki	u s O		tories .0	Re-a raisa Grou 3 Re-a raisa Year 2017					
Mu ti e- ₀ fami Units	W	es <mark>0</mark>	Garage	quare Ft <mark>0</mark>		Origina Construction Y	éar	Weighted	Year	
	Septic Ta	nks <mark>0</mark>	Attached	/ Detached						
	uildings So									
Res	sidence So	q Ft <mark>0</mark>								
Improvement List Bas	sement So	q Ft <mark>0</mark>	Baseme	ent						
Finished E	Basement	SF 0	Bedro	oms / Baths <mark>0 /</mark>						
				.00						
Assesse	d Valuat	tion				Taxal	ble Valuat	ion		
Assessed Values	2018-19	<u>2017-18</u>	2016-17			Taxable Values	<u>2018-19</u>	2017-18	2016-17	
Land	3,850	3,850	3,850			Land	11,000	11,000	11,000	
Improvements	0	0	0			Improvements	0	0	0	
Personal Property	0	0	0			Personal Property	0	0	0	
Ag Land	0	0	0			Ag Land	0	0	0	
Exemptions	0	0	0			Exemptions	0	0	0	
Net Assessed Value	3,850	3,850	3,850			Net Taxable Value	11,000	11,000	11,000	
Increased (New) Values	3					Increased (New) Value	es			
	0	0	0			Land	0	0	0	
Land										
Land Improvements Personal Property	0	0	0 0			Improvements Personal Property	0	0	0	

Back to Search List



Personal Property Sales Data

Secured Tax Inquiry Recorder Search



Back to Search List

STATE OF NEVADA DEPARTMENT OF TAXATION DIVISION OF LOCAL GOVERNMENT SERVICIES 1550 E. COLLEGE PARKWAY CARSON CITY, NEVADA 89706

2017-18 UNSECURED MINING PROPERTY AD VALOREM ASSESSMENT

Pursuant to NRS 362.100(1)(b) The Department of Taxation has appraised and assessed the property described below for ad valorem tax purposes. The values below must be place on the 2017-18 Unsecured Tax Roll for billing and collection purposes. The assessor should enter the amounts allowed for pollution control devices on the 2017-18 Unsecured Tax Roll as exempt value.

- COUNTY: CH
- COMPANY: CYRQ ENERGY, INC.
 - SITE: SODA LAKE GEOTHERMAL
- PIN NUMBER: 1928-07-002

		VALUE	VALUE
NEW REAL PROPERTY	\$	1,192,758	417,465
EXISTING REAL PROPERTY		1,568,176	548,862
REAL PROPERTY POLLUTION CONTROL	_	0	0
NET REAL PROPERTY	\$	2,760,934	966,327
NEW PERSONAL PROPERTY	\$	420,203	147,071
EXISTING PERSONAL PROPERTY		7,096,296	2,483,704
PERSONAL PROPERTY POLLUTION CONTROL		506,348	177,222
NET PERSONAL PROPERTY	\$	7,010,151	2,453,553
TOTAL NEW PROPERTY	\$	1,612,961	564,536
TOTAL EXISTING PROPERTY		8,664,472	3,032,566
TOTAL POLLUTION CONTROL		506,348	177,222
NET TOTAL PROPERTY	\$	9,771,085	3,419,880

	TAXABLE VALUE	ASSESSED VALUE)
*Amount of Total Value which is Eligble for the Renewable Energy Abatement	\$ 0		0

Any appeal of the value assessed by the Department must be submitted directly to the State Board of Equalization.

TAXABLE ASSESSED

		MAKE REMITIANCE PAYA	BLE TO:	
Denise L. Mo Churchill Co	ndhink-Felton unty Assessor	CHURCHILL CO CI	ERK/TREA	SURER STE 110
155 N. Taylo Fallon, NV	www.churchillcounty.or			
775-423-6584		rg CATION / DESCRIPTION		
ACCOUNT NO.	OWNER/LOCATION	CATION / DESCRIPTION	DISTRICT	DATE
MM 007054	CYRQ ENERGY INC			
		FORMERLY MAGMA	2.0	12/01/17
	Q ENERGY INC		and the set of the second set of	PERTY TAXES
	N: JOHN PERRY CFO SODA LAKE RD	F	OR FISCA	L YEAR:
	LON NV 89406-5369		2017-	-18
ASSE	SSED VALUATION	DISTRIBUTION	OF TAX AMO	UNTS
DESCRIPTIO		TAXING ENTITY	RATE	AMOUNT
Land	24,658 1 Property 2,630,775	General Co	0.8619	29,108.06
Improve	ments 966,327 ons 177,222-	Social Svcs Ag Extension Public Librar	0.0500 0.0200	1,688.60 675.44
	ax Abatement amount	Cap Imp Fund	0.0500	2,164.79 1,688.60
is 2,929.	88	Tax Act 1991 Senior Center	0.0219 0.0600	739.61 1,004.12
Account Type	: MINING EQUIP	Hosp Care MVA Fire Equip	$0.0150 \\ 0.0300$	506.58
Parcel #: 00	_	Fire Equip Mosq & Weed School Dist School Debt	0.0800 0.7500 0.5500	2,701.76
	5 571 65	St of Nevada Youth Svcs	0.1700	18,574.59 5,741.23
100 Creation of the call of th		Ind Med Care CWS District	0.0600	1,688.60 2,026.32 1,013.16
		Ad Valorem Tax Total		
		OSF STREET CA	1.11	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
		The second second second	15 200	
TOTAL ASSESSED				
	5,444,550			95,683.80
a 10	ment received later than 0% penalty. Please print ase advise the Assessor' ASE MAKE REMITTANCE PAYA	your Account Number	ate will b on your c	e subject to heck.
PLEA	ASE MAKE REMITTANCE PAYA	BLE TO: CHURCHILL COU	or addres JNTY CLERK	s changes. /TREASURER
		+	1	
OWNER FISCAL YEAR		OWNER FISCAL YEAR		
ACCOUNT NO.		ACCOUNT NO.		
DUE DATE	AMOUNT DUE	DUE DATE	AMOUNT DUE	
		XXXXXXXXXXXX		
Ath Installes ant				1251
4th Installment		3rd Installment		
OWNER		OWNER CYRO ENERGY	INC	
FISCAL YEAR ACCOUNT NO.		FISCAL YEAR 2017-18 ACCOUNT NO. MM 007054		
DUE DATE	AMOUNT DUE		AMOUNT DUE	95,683.80
XXXXXXXXXXXX		Jan 1, 2018	a dette	99,009.00
2nd Installment		1st Installment		
DETACH AND MAIL THIS STUB	20mmunaruanitanitanitani	DETACH AND MAIL THIS STUB		



Attachment 7 Company Website

https://www.cyrqenergy.com/



Attachment 8 Business License

SECRETARY OF STATE



NEVADA STATE BUSINESS LICENSE

AMOR IX, LLC

Nevada Business Identification # NV19881022354

Expiration Date: August 31, 2019

In accordance with Title 7 of Nevada Revised Statutes, pursuant to proper application duly filed and payment of appropriate prescribed fees, the above named is hereby granted a Nevada State Business License for business activities conducted within the State of Nevada.

Valid until the expiration date listed unless suspended, revoked or cancelled in accordance with the provisions in Nevada Revised Statutes. License is not transferable and is not in lieu of any local business license, permit or registration.



IN WITNESS WHEREOF, I have hereunto set my hand and affixed the Great Seal of State, at my office on June 14, 2018

Barbara K. Cegerste

Barbara K. Cegavske Secretary of State

You may verify this license at www.nvsos.gov under the Nevada Business Search.

License must be cancelled on or before its expiration date if business activity ceases. Failure to do so will result in late fees or penalties which by law <u>cannot</u> be waived.





Attachment 9 Facility Information

State of Nevada Renewable Energy Tax Abatement Application

AFN:
Facility Information
Date of Submittal to GOE:
Type of Incentives (Please check all that the company is applying for on this application)
X Sales & Use Tax Abatement X Property Tax Abatement
Company Information (Legal name of company under which business will be transacted in
Company Name: Amor IX, LLC
Department of Taxation's Tax Payer ID number: 1007968940-001
Federal Employer ID number (FEIN, EIN or FID): 26-3896834
NAICS Code: 221116
Description of Company's Nevada Operations: AMOR IX, LLC is the owner and operator of the Soda Lake Geothermal Power Plant just outside of Fallon, NV. AMOR IX, LLC plans to repower Soda Lake Geothermal by adding a new generating unit and decomissioning the existing Soda Lake I and II plants. Cyrq Energy, Inc. (Ultimate Parent Company of AMOR IX, LLC) also owns the Patua Geothermal and Solar Power Plants in nearby Hazen, NV as well as the Nevada Geothermal Utility Company in Reno, NV.
Percentage of Company's Market Inside Nevada: 0%
Mailing Address: 136 S Main St STE 600
City: Salt Lake City, UT 84101
Phone: 801.875.4200
APN: 009-371-85, 009-391-17, 009-391-14
Taxation District where facility is located: Churchill County District No. 2.0
Nevada Facility
Type of Facility (please check all that are relevant to the facility)
X Geothermal Process Heat from Solar Energy Solar PV
Solar Thermal Wind Biomass Waterpower Fuel Cells
X Transmission that is interconnected to a renewable energy or geothermal Transmission that contributes to the capability of the electrical grid to accommodate and transmit electricity produced from Nevada renewable Name Plate Production Capacity of the Facility: 23.87 MW (Gross)

State of Nevada Renewable Energy Tax Abatement Application AFN:

AFN:	
Net Output Production Capacity of the Facility in MW: 21.2 MW (Facility Net Power excluding Wellfield)	
Annual Net Production Capacity of the Facility in MWh (or other appropriate unit): 145,954 MWH	
Estimated total capital investment: \$ 37,240,789.00	
Percent of total estimated capital investment expended in Nevada:	100%
Anticipated date or time range for the start of construction: March 1, 2019	
Anticipated date for the Commerical Operation Date (COD) of the facility: December 31, 2019	
Construction period (in months). Note: time period muct match payroll calculations	10
Address of the Real Property for the Generation Facility: 5500 Soda Lake Road	
City: Fallon, NV 89406	

Size of the total Facility Land (acre):

1,420

Are yo	Are you required to file any paper work with the PUC and/or FERC? Yes, but to FERC only.						
lf yes,	Purpose of the Filing with PUC:	Filing Date OR Anticipated filing Date:					
lf yes,	Purpose of the Filing with FERC: Market-based Rate Authorization	Filing Date OR Anticipated filing Date: March 1, 2019					

L	List All the county(s), Cities, and Towns where the facility will be					
1	Churchill County, Nevada					
2						
3						
4						
5						
6						
7						
8						
9						

State of Nevada Renewable Energy Tax Abatement Application AFN:

List of Required Permits or Authorizations for the Proposed Facility

	Permit or Authorization Title	Issuing Agency	Project Circumstance Requiring Permit or Authorization	Steps to Obtain Permit	Application Date	Approval Date or Expected Approval Date
Endora	I Permits or Authoriza	tione				
Teuera	Radio Station Authorization	Feder Communications Commission	Per Federal regulatory framework	Acquired. Expires 09/16/2024.	7/31/2014	7/31/2014
	Exempt Wholesale Generator Status and Market-based Rate Authority	Federal Energy Regulatory Commission	Per Federal regulatory framework	Self-Certification for EWG Status, and file Docket with FERC for MBR Authority		
I. State (of Nevada Permits or A	Authorizations				
	Amendment to Class II Air Quality Operating Permit - AP4911-0464.04, (FIN A0548)	NV DEP BAPC	Local State and Federal regulatory framework	Acquired. Revised 05/08/2017. Expires 10/15/2019.	12/1/2014	12/18/2014
	Stormwater Construction Permit/SWPPP	NV DEP BWC	Local State and Federal regulatory framework	Analysis, documentation, and application necessary to fulfill regulatory requirements.		
	Chemical Accident Prevention Program (CAPP) Permit to Construct	NV DEP DCNR	Per regulations and to ensure proper handling of hazardous chemicals and explosives.	The EPC Contractor will arrange and perform the Process Hazard Analysis (PHA) as required by CAPP, and will assist in the permit to construct process. Capp Permitting and Construction are closely coordinated.		
	Underground Injection Control Permit (UIC Permit UNEV89037)	NV DEP DCNR	Local State and Federal regulatory framework	Acquired. Expires 05/05/2019		5/8/2014
	Goethermal Statewide Drilling Surety Bond RLB0015954 (\$50,000)	NV DOM	Local State and Federal regulatory framework	Acquired. Effective 02/12/2015	2/12/2015	2/12/2015
	Boiler Pressure Vessel Building Permits	NV DBI MCS	Local State and Federal regulatory framework	Analysis, documentation, and application necessary to fulfill regulatory requirements.		
	Boiler Pressure Vessel Operating Permits	NV DBI MCS	Local State and Federal regulatory framework	Analysis, documentation, and application necessary to fulfill regulatory requirements.		
	Hazardous Materials Permit	NV State Fire Marshall	Local State and Federal regulatory framework	Analysis, documentation, and application necessary to fulfill regulatory requirements.		
	Business License (NV19881022354)	State of Nevada	Local State and Federal regulatory framework	Acquired. Expires 08/31/2018		8/7/2017
II. Coun	ty Permits or Authoriz	ations				
	Special Use Permit	Churchill County	Local State and Federal regulatory framework	Analysis, documentation, and application necessary to fulfill regulatory requirements.	9/7/2016	
	Building Permit and Grading Permit	Churchill County	Local State and Federal regulatory framework	Analysis, documentation, and application necessary to fulfill regulatory requirements.		
	Business License (No. 12594)	Churchill County	Local State and Federal Regulatory framework	Acquired. Expires 12/31/2017		12/16/2016
	Permits or Authorization	ons				
	1					



Attachment 10 Employment Information

State of Nevada Renewable Energy Tax Abatement Application

AFN:

Employment Information

Employment

New Operations or Expansion		
	Full Time	Part Time
Number of anticipated construction employees who will be employed during the entire construction phase?	71	18
Number of anticipated construction employees who will be employed during the entire construction phase that will be Nevada Residents?	36	10
Average anticipated hourly wage of construction employees, excluding management and administrative employees:	38.66	38.66
Number of anticipated construction employees who will be employed during the second-quarter of construction?	62	15
Percentage of anticipated second-quarter construction employees who will be Nevada Residents?	80%	33%
Number of anticipated second-quarter construction employees who will be Nevada Residents?	50	5
PERMANENT EMPLOYEES		
Number of anticipated permanent employees who will be employed as of the end of its first fourth-quarter of new operations or expansion?	12	N/A
Average anticipated hourly wage of permanent employees, excluding management and administrative employees:	34.85	N/A
Number of permanent employees who were employed prior to the expansion?	12	N/A
Average hourly wage of current permanent employees, excluding managements and administrative employees	34.85	N/A

Employee Benefit Program for Construction Employees

Health insurance for construction employees and an option for dependents must be offered upon employment

List Benefits Included (medical, dental, vision, flex spending account, etc): Determined by Vendor - We will require that vendors meet the qualificiations of Chapter 701A and applicable regulations or their bids will not be accepted.						
Name of Insurer: Determined by Vendor						
Cost of Total Benefit Package:	600,000	Cost of Health Insurance for Construction Employees:	Estimated at 8% of salaries			

State of Nevada Renewable Energy Tax Abatement Application AFN:

Construction Employee Schedule

List all anticipated construction employees and associated wages for all persons who will be working on the construction of the facility during the entire construction period. *Please provide the formula utilized to arrive at the numbers below**

FULL TIME EMPLOYEES

		(a)	(b)	(c) = (a)+(b)	(e) = (c) x (d)	(d)
					Total Hourly	
		# of Nevada	# of Non-Nevada	Total # of	Wage per	Average Hourly
#	Job Title	Employees	Employees	Employees	category (\$)	Wage (\$)

						(f) =Σ(e) / Σ
	TOTAL	55	16	71	\$2,744.86	\$38.66
Ormat		4	0	4	\$154.64	\$38.66
Insulation		2	6	8	\$309.28	\$38.66
Fencing		2	0	2	\$77.32	\$38.66
Buildings		4	0	4	\$154.64	\$38.66
Electrical Employees		10	2	12	\$463.92	\$38.66
Mechancial Employees		20	8	28	\$1,082.48	\$38.66
Fire Construction Employees		3	0	3	\$115.98	\$38.66
Civil Construction Employees		10	0	10	\$386.60	\$38.66

Formula: $\sum_{1}^{n} (N_n x H x A W)$

a. "N" Number of construction employees (during a month of construction).

b. "n" number of the construction month. (total anticipated 10 months)

c. "H" Number of work hours a month.

d. "AW" Average Wage 38.66\$/hr.

State of Nevada Renewable Energy Tax Abatement Application AFN:

Second Quarter Construction Employee Schedule

List all anticipated construction employees and associated wages for all persons who will be working on the construction of the facility during the second quarter of construction. *Please provide the formula utilized to arrive at the numbers below**

FULL TIME EMPLOYEES

_			(a)	(b)	(c) = (a)+(b)	(e) = (c) x (d)	(d)
ſ						Total Hourly	Average
			# of Nevada	# of Non-Nevada	Total # of	Wage per job	Hourly Wage
	#	Job Title	Employees	Employees	Employees	title (\$)	(\$)

					(f) =Σ(e) /
TOTAL	55		22 77	\$2,976.82	\$38.6
Ormat Employees	4	0	4	\$154.64	\$38.66
Fencing Construction Employees	3	0	3	\$115.98	\$38.66
Building Construction Employees	5	0	5	\$193.30	\$38.66
Electrical Construction Employees	8	7	15	\$579.90	\$38.66
Mechanical Construciton Employees	25	10	35	\$1,353.10	\$38.60
Civil Construction Employees	10	5	15	\$579.90	\$38.66

TOTAL CONSTRUCTION PAYROLL	\$1,252,584
----------------------------	-------------

State of Nevada Renewable Energy Tax Abatement Application AFN: Permanent Employee Schedule

List all anticipated permanent employees who will be employed by the Nevada Facility as of the end of its first fourth-quarter of new operations or expansion and the employment per job title will continue next 20 years

FULL TIME EMPLOYEES

		(c)	(f) = Σ(e) / Σ(c)
		# of	Average Hourly
#	Job Title	Employees	

1	Management and Administrative Employees	2	\$45.22
2	Permanent Employees, excluding Management	10	\$34.85
	and Administrative Employees		
	TOTAL	12	

TOTAL	L PAYR	

746,591

Payroll Formula					
Total Annual Payroll	\$746,591.00				
Employees	10.00				
Hours Per Week	41.20				
Weeks	52.00				
Average Hourly Wage	\$ 34.85				



Attachment 11 Supplemental Information

State of Nevada Renewable Energy Tax Abatement Application AFN:

Please respond to each question. Answers to the questions will assist Department of Taxation staff in determining whether the facility should be locally or centrally assessed. Other questions will assist staff in understanding whether the reported replacement costs capture all aspects of taxable value.

1) Will you have a possessory interest in any governmentally owned property for this facility? Please describe if yes.

Yes. BLM Geothermal Resource Leases N-11737 and N-53371

2) Will the facility, including generation, transmission, or distribution cross state or county boundaries? If yes, please describe.

Yes. The Soda Lake Facility will interconnect at Ragtown 63kV Substation located in Churchill County, Nevada. The power will then be wheeled through the NV Energy Balancing Authority and delivered at Gonder Substation, located in White Pine County, Nevada. Rocky Mountain Power will facilitate transmission of energy across state boundaries into Utah.

3) Is the facility owned by a subsidiary of a company that is interstate or intercounty in nature? Name and location of the subsidiary company, if yes.

Yes. Cyrq Power Systems, LLC, a Delaware entity, manages Amor IX, LLC. Cyrq Power Systems is a subsidiary of Cyrq Energy, Inc., which is also a Delaware entity, and both operate out of the State of Utah. Amor IX, LLC's (a subsidiary of Cyrq Power Systems, LLC) primary place of business is in the State of Nevada.

4) At what physical point is the ownership of energy transferred? Describe the location and nature of the connection to the transmission grid.

The Facility interconnects to the transmission grid at Ragtown Substation in Churchill County via a Large Generator Interconnection Agreement with NV Energy. Power is wheeled through the NV Energy Transmission System under a Transmission Service Agreement. Ownership of the energy is transferred at the Gonder Substation where Rocky Mountain Power takes control of energy delivery into the state of Utah.

5) Will the facility be eligible for other abatements or exemptions such as pollution control exemptions? Please describe if yes

No

6) Has your company applied and/or been approved for any abatements or exemptions for this facility or any other facility by the State of Nevada and/or local governments? If yes, list the abatements awarded, name and location of the project, name of the awarder, date of approval, amounts and status of the accounts

No, AMOR IX, LLC previously has not applied for state or local abatements in the State of Nevada and/or local governments. **7) Has your company applied for, or planning to apply for, an exempt wholesale generator designation as defined in 15 U.S.C 79z-5A?**

Yes

8) If an EIS or EA has been performed, please supply the ROD number.

Yes (NV-030-9006) Finding of No Significant Impact

9) Has an appraisal been performed on any portion of this land or project?

No

10) Has a Power Purchase Agreement been executed?

Yes, but is still subject to the State of Utah Public Utility Commission Approval



Attachment 12 Tax Schedules

State of Nevada Renewable Energy Tax Abatements Application AFN:

Summary Report Schedules 1 through 8

Company: AMOR IX, LLC

Division: Soda Lake Geothermal Power Plant

Line No.	Schedule	Total Estimated RCNLD or Transaction Cost	Department Use Only
1	Sch. 1 Personal Property - Property Tax - Total from Col. J. *	-	
2	Sch. 2 Real Property - Improvements - Total from Col. F. *	37,240,789	
3	Sch. 3 Real Property - Land - Total from Col. I	95,000	
4	Sch. 4 Operating Leases - Total from Col. F *	170,958	
5	Sch. 5 Contributions in Aid of Construction - Total from Col. F	-	
6	Sch. 6 First Year Estimated Sales & Use Tax - Total from Col. J	1,990,580	
7	Sch. 7 Second Year Estimated Sales & Use Tax - Total from Col. J	87,741	
8	Sch. 8 Third Year Estimated Sales & Use Tax - Total from Col. J	69,035	

* The final determination of the classification of property as real or personal is made by the county assessor for locally-assessed property or by the Department of Taxation for centrally-assessed property. Placement of property on these sheets of the application is made for purposes of this fiscal note only and is not determinative of the final classification of property by the appropriate taxing official.

State of Nevada Renewable Energy Tax Abatements Application AFN:

Company Name: AMOR IX, LLC

Division: Soda Lake Geothermal Power Plant

Property Tax: Personal Property Schedule 1

Instructions:

(1) List each item of personal property subject to property tax in Col A. Pursuant to NRS 361.030, personal property includes stocks of goods on hand; any vehicle not included in the definition of vehicle in NRS 371.020; all machines and machinery, all works and improvements, and all property of whatever kind or nature not included in the term "real estate" as that term is defined in NRS 361.035.

(2) For each item in Col. A, complete the requested information in Col. B and Col. D (if applicable), Col. C and Col. D through Col. J.

(3) The total estimated cost reported in Col. H should include estimated or actual costs of installation and costs of transportation per NAC 361.1351 and NAC 361.1355. Costs of installation include the costs of direct labor, direct overhead and the capitalized expense of interest or imputed charges for interest which are necessary to make the property operational.

(4) Use the Personal Property Manual published by the Department of Taxation to determine the Cost Less Depreciation in Column (J). Select the Life Schedule that is closest to the estimated life of the personal property listed in Col. I. See http://tax.state.nv.us. Then select: Publications/Locally Assessed Properties/Personal Property Manual.

(5) Attach additional sheets as necessary.

Α	В	С	D	E	Н	I	J
Personal Property Itemized Description	G/L Account No. (if applicable)	Purchased by Facility Owner (FO) Contractor (C) Subcontractor (SC)	Date Purchased (if applicable)	Date Received or Estimated Date of Receipt in Nevada	Estimated Total Acquisition Cost	Estimated Life of Personal Property	Estimated Acquisition Cost Less Depreciation
Included under Schedule 2 - Real Property,	/Improvements						
Grand Total							

State of Nevada Renewable Energy Tax Abatements Application AFN:

Company Name: AMOR IX, LLCProperty Tax: Real Property ImprovementsDivision: Soda Lake Geothermal Power PlantSchedule 2

Instructions:

(1) List each item of real property improvements subject to property tax in Col A. Pursuant to NRS 361.035, real property includes all houses, buildings, fences, ditches, structures, erections, railroads, toll roads and bridges, or other improvements built or erected upon any land, whether such land is private property or public property; as well as mobile or manufactured

(2) For each item in Col. A, complete the requested information in Col. B (if applicable), and Col. C through Col. F.

(3) The total estimated cost reported in Col. F should include estimated or actual costs of labor (do not include construction or operational employee totals from previous tab), materials, supervision, contractors' profit and overhead, architects' plans and specifications, engineering plans, building permits, site preparation costs, sales taxes and insurance; costs of buying or assembling land such as escrow fees, legal fees, right of way costs, demolition, storm drains, rough grading or other land improvement costs, yard improvements including septic systems, signs, landscaping, paving, walls, yard lighting; off-site

(4) Use Schedule 3 to report land; Schedule 4 to report operating leases; and Schedule 5 to report contributions in aid of

Α	В	C	F
Real Property Improvements Itemized Description	G/L Account No. (if applicable)	Estimated Date of Completion	Estimated Total Construction Cost
Project Management		July, 2019	\$ 1,250,000
Project Engineering		July, 2019	\$ 435,000
Project Construction and Fees		July, 2019	\$ 8,900,000
OEC Unit		July, 2019	\$ 19,700,000
Geothermal System		July, 2019	\$ 500,000
Fire Fighting System		July, 2019	\$ 350,000
Auxillary System		July, 2019	\$ 500,000
Electrical System		July, 2019	\$ 1,200,000
Transportation		July, 2019	\$ 3,000,000
Building		July, 2019	\$ 150,000
Construction Management		July, 2019	\$ 200,000
Switchyard Upgrades		July, 2019	\$ 455,789
Pipeline Upgrades		July, 2019	\$ 600,000
Grand Total			\$ 37,240,789

(5) Attach additional sheets as necessary.
Company: AMOR IX, LLC

Property Tax: Real Property Land Schedule 3

Division: Soda Lake Geothermal Power Plant

Show the requested data for **all land**, owned or leased, in Nevada.

Α	В	С	D	Н		F	G	Н	
		Where Situate	d			Owned (O)	G/L Account Number	Purchase	
Line #	County	City or Town	Tax District	Land (acre), Date Acquired	Assessor's Parcel Number (APN)	Leased (L) Rented (Rtd)	(if applicable)	Price (if applicable)	Assessor's Taxable Value
1	Churchill		2.0	TCID - S33 T20N R28E, 640 acres, 05/01/1978	009-371-85	L	N/A	N/A	\$65,000.00
2	Churchill		2.0	S28 T20N R28E N2, 320 acres, 10/01/1975 (BLM N-11737)	009-391-17	L	N/A	N/A	\$9,500.00
3	Churchill		2.0	S28 T20N R28E S2, 320 acres, 10/01/1975 (BLM N-53371)	009-391-17	L	N/A	N/A	\$9,500.00
4	Churchill		2.0	T20N R28E S29 S2SE4, NE4SE4, E2NW4SE4, 140 acres, 03/01/1979	009-391-14	L	N/A	N/A	\$11,000.00
5									
6									
7									
8									
9									
10									
11									
12	Grand Total								\$95,000

Property Tax: Operating Leases Schedule 4

Company Name: AMOR IX, LLC

Division: Soda Lake Geothermal Power Plant

Instructions:

(1) List each operating lease for real or personal property. Designate whether the lease is for real or personal property in Col. C.

(2) For each item in Col. A, complete the requested information in Col. B (if applicable), and Col. C through Col. F.

(3) The total estimated cost reported in Col. E and Col. F should contain the costs appropriate to real or personal property. For definitions, please refer to Schedule 1 for personal property and Schedule 2 for Improvements.

(4) Report the Annual Lease Payment in Col. G; the term of the lease in Col. H; and any residual value at the end of the lease term in Col. I.

(5) Attach additional sheets as necessary.

Α	В	С	E	F	G	Н	I
Operating Lease Itemized Description	G/L Account No. (if applicable)	Real or Personal Property?	Lessor's Replacement Cost Per Unit	Estimated Total Replacement Cost	Annual Lease payment	Lease Years Remaining	Residual Value
Truckee-Carson Irrigation District (TCID) - Site							
Sublease II - S33 T20N R28E, 640 acres,							
05/01/1978	N/A	Real			\$162,000	Unit Agreement	
BLM N-11737 - S28 T20N R28E N2, 320							
acres, 10/01/1975	N/A	Real			\$2,238	Unit Agreement	
BLM N-53371 -S28 T20N R28E S2, 320 acres,							
10/01/1975	N/A	Real			\$1,120	Unit Agreement	
Powelson/Henderson - T20N R28E S29 S2SE4, NE4SE4, E2NW4SE4, 140 acres,							
03/01/1979	N/A	Real			\$5,600	Unit Agreement	
							ļ
Grand Total							

Company Name: AMOR IX, LLC Division: Soda Lake Geothermal Power Plant

Instructions:

Property Tax: Contributions in Aid of Construction Schedule 5

(1) List all contributions in aid of construction (CIAC). CIAC is defined in NAC 361.260 as property which has been contributed to a utility by a prospective customer or which has been constructed by the utility and paid for by the prospective customer for which no reimbursement is required to be made by the utility to the prospective customer as a prerequisite to obtaining service.

(2) For each item in Col. A, complete the requested information in Col. B (if applicable), and Col. C through Col. F.

(3) The total estimated cost reported in Col. E and Col. F should contain the costs appropriate to real or personal property. For definitions, please refer to Schedule 1 for personal property and Schedule 2 for Improvements.

(4) Attach additional sheets as necessary.

Α	В	С	D	E	F
					Estimated
					Total
Contributions in Aid of Construction (CIAC)	G/L Account No.	Real or Personal		Replacement	Replacement
Itemized Description	(if applicable)	Property?	Number of Units	Cost Per Unit	Cost
N/A					
Crond Total					l
Grand Total					

Company Name: AMOR IX, LLC

Division: Soda Lake Geothermal Power Plant

Sales and Use Tax First Year of Eligible Abatement Schedule 6

Instructions:

(1) Column A: List each item of personal property or materials and supplies subject to sales and use tax (please include leases. Refer to NRS Chapter 372 for taxable events.

(2) Column B: For each item in column A, list applicable account nomber.

(3) Column C: List the Facility Owner, Contractor or Subcontractor that will be purchasing the personal property or materials and supplies subject to sales and use tax.

(4) Column D: List the date the personal property or materials and supplies were purchased.

(5) Column E: List the date that possession of the personal property or materials and supplies will be taken.

(6) Column F: List the cost of the personal property or materials and supplies.

(7) Column G: List the county where possession will be taken and the applicable sales tax rate of that county. Find the appropriate sales/use tax rate on the Department of Taxation's website at http://tax.state.nv.us. Then scroll to "Quick Links" and select "Sales/Use Tax Rate Map".

(8) Column H: Multiply Column F by the Sales Tax Rate in Column G.

(9) Attach additional sheets as necessary.

А	В	C	D	E	F	G	Н
Personal Property or Materials and	G/L Account No. (if	Purchased by Facility Owner (FO) Contractor (C)	Date	Date of	Total Transaction	County and Applicable Sales Tax	Estimated Sales Tax Paid or to
Supplies Itemized Description	applicable)	Subcontractor (SC)	Purchased		Cost	Rate	be Paid
OEC Unit		C/SC			\$ 19,700,000	0.076	\$ 1,497,200
Geothermal System		C/SC			\$ 500,000	0.076	
Fire Fighting System		C/SC			\$ 350,000	0.076	\$ 26,600
Auxillary System		C/SC			\$ 500,000	0.076	\$ 38,000
Electrical System		C/SC			\$ 1,200,000	0.076	\$ 91,200
Transportation		C/SC			\$ 3,000,000	0.076	\$ 228,000
Building		C/SC			\$ 150,000	0.076	\$ 11,400
Switchyard Upgrades		FO/SC			\$ 341,842	0.076	\$ 25,980
Pipeline Upgrades		FO/SC			\$ 450,000	0.076	\$ 34,200
Grand Total							\$ 1,990,580

Company Name: AMOR IX, LLC

Division: Soda Lake Geothermal Power Plant

Sales and Use Tax Second Year of Eligible Abatement Schedule 7

Instructions:

(1) Column A: List each item of personal property or materials and supplies subject to sales and use tax (please include leases. Refer to NRS Chapter 372 for taxable events.

(2) Column B: For each item in column A, list applicable account nomber.

(3) Column C: List the Facility Owner, Contractor or Subcontractor that will be purchasing the personal property or materials and supplies subject to sales and use tax.

(4) Column D: List the date the personal property or materials and supplies were purchased.

(5) Column E: List the date that possession of the personal property or materials and supplies will be taken.

(6) Column F: List the cost of the personal property or materials and supplies.

(7) Column G: List the county where possession will be taken and the applicable sales tax rate of that county. Find the appropriate sales/use tax rate on the Department of Taxation's website at http://tax.state.nv.us. Then scroll to "Quick Links" and select "Sales/Use Tax Rate Map".

(8) Column H: Multiply Column F by the Sales Tax Rate in Column G.

(9) Attach additional sheets as necessary.

А	В	С	D	E	F	G	Н
Personal Property or Materials and Supplies Itemized Description	G/L Account No. (if applicable)	Purchased by Facility Owner (FO) Contractor (C) Subcontractor (SC)	Date Purchased	Date of Possession	Total Transaction Cost	County and Applicable Sales Tax Rate	Estimated Sales Tax Paid or to be Paid
OEC Unit	appricable)	C/SC	i aronacoa	1 0000001011		0.076	
Geothermal System		C/SC				0.076	
Fire Fighting System		C/SC				0.076	
Auxillary System		C/SC				0.076	
Electrical System		C/SC				0.076	
Transportation		C/SC				0.076	
Building		C/SC				0.076	
Switchyard Upgrades		FO/SC			\$ 113,947	0.076	\$ 8,660
Pipeline Upgrades		FO/SC			\$ 150,000	0.076	\$ 11,400
O&M - Materials & Supplies		FO			\$ 747,912	0.076	\$ 56,841
O&M - Repair Parts & Maintenance		FO			\$ 120,028	0.076	\$ 9,122
O&M - Other		FO			\$ 22,606	0.076	
Grand Total							\$ 87,741

Company Name: AMOR IX, LLC

Division: Soda Lake Geothermal Power Plant

Sales and Use Tax Third Year of Eligible Abatement Schedule 8

.

.. .

Instructions:

(1) Column A: List each item of personal property or materials and supplies subject to sales and use tax (please include leases. Refer to NRS Chapter 372 for taxable events.

(2) Column B: For each item in column A, list applicable account nomber.

(3) Column C: List the Facility Owner, Contractor or Subcontractor that will be purchasing the personal property or materials and supplies subject to sales and use tax.

(4) Column D: List the date the personal property or materials and supplies were purchased.

(5) Column E: List the date that possession of the personal property or materials and supplies will be taken.

(6) Column F: List the cost of the personal property or materials and supplies.

http://tax.state.nv.us. Then scroll to "Quick Links" and select "Sales/Use Tax Rate Map".

(7) Column H: Multiply Column F by the Sales Tax Rate in Column G.

(8) Attach additional sheets as necessary.

А	В	С	D	E	F	G	Н
Personal Property or Materials and Supplies Itemized Description	G/L Account No. (if applicable)	Purchased by Facility Owner (FO) Contractor (C) Subcontractor (SC)	Date Purchased	Date of Possession	Total Transaction Cost	County and Applicable Sales Tax Rate	Estimated Sales Tax Paid or to be Paid
O&M - Materials & Supplies		FO			\$ 762,870	0.076	\$ 57,978
O&M - Repair Parts & Maintenance		FO			\$ 122,428	0.076	\$ 9,305
O&M - Other		FO			\$ 23,058	0.076	\$ 1,752
Grand Total							\$ 69,035



Attachment 13 Contractors & Subcontractors

NOTE: Project contractors, subcontractors, and other entities including owner that will be purchasing goods and equipment for the construction of the Facility are entitled to claim or receive the sales and use tax abatement

Vendor	Paul Graham Drilling and Service Company
Tax ID	1019844051-001
Contact	Jane Stewart
Mailing Address	PO Box 1027
	Rio Vista, CA 94571
E-Mail	jane@paulgrahamdrilling.com
Vendor	Tholl Fence, Inc.
Tax ID	1002032180
Contact	Shawna Downs
Mailing Address	800 Glendale Ave
	Sparks, NV 89431
E-Mail	snawna@unoilience.com

Contractors and Subcontractors List

Vendor	Applied High Voltage
Tax ID	61-1704078
Contact	Sheree Quarles
Mailing Address	2034 Hamilton Place Blvd Ste400 Chattanooga, TN 37421
E-Mail	<u>ahvtax@emjtax.com</u>

Vendor	Geo Drilling Fluids, Inc.
Tax ID	06-1047211
Contact	Brad Bush
Maning Address	1431 Union Ave
	Bakersfield, CA 93305
E-Mail	bebush@sbcglobal.net

Vendor	Probst Electric Inc.
Tax ID	56-2477123
Contact	Timmothy Titus
Mailing Address	875 S. 600 W.
	Heber City, UT 84032
E-Mail	timothy@probstelectric.com

Vendor	Glacier Construction Inc.
Tax ID	20-3334106
Contact	Julie or Rhonda Major
Mailing Address	4606 Beacon Wy
	Fallon, NV 89406
E-Mail	pcbsbooks@yahoo.com
	-

Vendor	The Best Company Inc. dba Colorado TBC
--------	--

AFN:		
Tax ID	1000215318-001	
Contact	Norman Clark or Shelby Cooper	
Mailing Address	2258 Reno Highway STE D	
	Fallon, NV 89406	
	nclark@coloradotbc.com or	
E-Mail	rcooper@coloradotbc.com	

Vendor	Oasis Air Conditioning
Tax ID	88-0148448
Contact	Summer Shuey
Mailing Address	1931 Grimes St.
	Fallon, NV 89406
E-Mail	summer@oasishvacnv.com

Vendor	Hammond Homes
Tax ID	20-8314936
Contact	Laurel Hammond
Mailing Address	1780 Lattin Rd
	Fallon, NV 89406
E-Mail	hammon5@phonewave.net

Vendor	Merit Electric Company
Tax ID	1001649427
Contact	Debbie Holbrook
Mailing Address	7785 White Fir St.
	Reno, NV 89523
E-Mail	dholbrook@meritelectricreno.com

Vendor	SimplexGrinnell LP
Tax ID	58-2608861
Contact	Matthew Rosecrans
Mailing Address	6952 Preston Ave Suite A
	Livermore, CA 94551
E-Mail	mrosecrans@simplexgrinnell.com



Attachment 14 Power Purchase Agreement

Rocky Mountain Power Exhibit RMP___(MPT-2) Page 1 of 82 Docket No. 18-035-08 Witness: Mark P. Tourangeau

RENEWABLE RESOURCE CONTRACT (SCHEDULE 32)

BETWEEN

AMOR IX, LLC

AND

PACIFICORP

TABLE OF CONTENTS

SECTION	1 DEFINITIONS, RULES OF INTERPRETATION 1
1.1	Defined Terms1
1.2	Rules of Interpretation11
SECTION	2 TERM
2.1	Term11
2.2	Project12
2.3	PacifiCorp's Right to Monitor
2.4	Tax Credits12
SECTION	3 REPRESENTATIONS AND WARRANTIES
3.1	Mutual Representations and Warranties13
3.2	Seller's Further Representations and Warranties13
3.3	PacifiCorp's Further Representations and Warranties15
3.4	No Other Representations or Warranties16
3.5	Continuing Nature of Representations and Warranties; Notice16
SECTION	4 DELIVERIES OF NET OUTPUT16
4.1	Purchase and Sale16
4.2	No Sales to Third Parties16
4.3	Title and Risk of Loss of Net Output17
4.4	Curtailment17
4.5	PacifiCorp as Merchant17
4.6	Title to RECs17
4.7	Capacity Rights17

TABLE OF CONTENTS (continued)

4	1.8	Third Party Sales of PECs	18
SECTIO	ON 5 C	CONTRACT PRICE; COSTS	18
5	5.1	Contract Price; Includes Capacity Rights	18
5	5.2	Costs and Charges	18
5	5.3	Station Service	18
5	5.4	Taxes	19
5	5.5	Costs of Ownership and Operation	19
5	5.6	Rates Not Subject to Review	19
SECTIO	ON 6 C	OPERATION AND CONTROL	19
ϵ	5.1	Standard of Facility Operation	19
6	5.2	Interconnection	20
6	5.3	Coordination with System	20
6	5.4	Outages.	21
6	5.5	Scheduling	22
6	5.6	Forecasting	22
6	5.7	Increase in Nameplate Capacity Rating; New Project Expansion or Development	23
6	5.8	Electronic Communications	23
6	5.9	Reports and Records.	23
6	5.10	Financial and Accounting Information	26
6	5.11	Output Guarantee.	26
6	5.12	Access Rights	27
6	5.13	Facility Images	27

TABLE OF CONTENTS (continued)

SECTION 7 FACILITY STATUS		
SECTION 8 SECURITY		
8.1	Project Development Security	
8.2	Default Security	
8.3	No Limitation	
SECTION 9	METERING	
9.1	Installation of Metering Equipment	
9.2	Metering	
9.3	Inspection, Testing, Repair and Replacement of Meters	
9.4	Metering Costs	
9.5	Meter Data	
9.6	WREGIS Metering	
SECTION 10	BILLINGS, COMPUTATIONS AND PAYMENTS	
10.1	Monthly Invoices	
10.2	Offsets	
10.3	Interest on Late Payments	
10.4	Disputed Amounts	
10.5	Audit Rights	
SECTION 11	DEFAULTS AND REMEDIES	
11.1	Defaults	
11.2	Remedies for Failure to Deliver/Receive	
11.3	Remedies for University Default	

TABLE OF CONTENTS (continued)

	11.4	Termination and Remedies	32
	11.5	Termination of Duty to Buy	33
	11.6	Termination Damages	33
	11.7	Duty/Right to Mitigate	33
	11.8	Security	34
	11.9	Cure by Lender or University.	34
	11.10	Cumulative Remedies	34
	11.11	Recoupment of Damages.	34
SECTI	ON 12	EARLY TERMINATION	35
SECTI	ON 13	INDEMNIFICATION AND LIABILITY	35
	13.1	Indemnities	35
SECTI	ON 14	INSURANCE	36
	14.1	Required Policies and Coverages	36
	14.2	Certificates of Insurance	
SECTI			36
SECTI		Certificates of Insurance	36 37
SECTI	ON 15 15.1	Certificates of Insurance	36 37 37
SECTI	ON 15 15.1	Certificates of Insurance FORCE MAJEURE	36 37 37 37
SECTI	ON 15 15.1 15.2	Certificates of Insurance	36 37 37 37 37
SECTI	ON 15 15.1 15.2 15.3	Certificates of Insurance	36 37 37 37 37 37
SECTI	ON 15 15.1 15.2 15.3 15.4	Certificates of Insurance	36 37 37 37 37 37 37

TABLE OF CONTENTS (continued)

SECTIC	ON 16	SEVERAL OBLIGATIONS	,
SECTIC	ON 17	CHOICE OF LAW	;
SECTIC) N 18	PARTIAL INVALIDITY	,
SECTIC) N 19	NON-WAIVER	,
SECTIC	ON 20	GOVERNMENTAL JURISDICTION AND AUTHORIZATIONS	,
SECTIC	ON 21	SUCCESSORS AND ASSIGNS)
2	21.1	Restriction on Assignments)
2	21.2	Permitted Assignments)
SECTIC	ON 22	ENTIRE AGREEMENT 40)
SECTIC	ON 23	NOTICES)
2	23.1	Addresses and Delivery Methods)
2	23.2	Changes of Address)
SECTIC	ON 24	CONFIDENTIALITY	!
2	24.1	Confidential Business Information)
2	24.2	Duty to Maintain Confidentiality43	;
2	24.3	PacifiCorp Regulatory Compliance	;
2	24.4	Irreparable Injury; Remedies	;
2	24.5	News Releases and Publicity	;
SECTIC	ON 25	DISAGREEMENTS 44	ŀ
2	25.1	Negotiations	ł
2	25.2	Mediation; Technical Expert	ŀ

TABLE OF CONTENTS

(continued)

	25.3	Place of Contract Formation; Choice of Forum	46
	25.4	Settlement Discussions	47
	25.5	Waiver of Jury Trial	47
SECT	ION 26	COMMISSION APPROVAL	47
SECT	ION 27	MATERIAL TERMS AND CONDITIONS – UNIVERSITY	48

EXHIBITS

Exhibit A	Expected Energy
Exhibit B	Description of Facility and Premises
Exhibit C	NERC Event Types
Exhibit D	Contract Price
Exhibit E	Permits
Exhibit F	Required Facility Documents
Exhibit G	Leases
Exhibit H	PacifiCorp's Initial Designated Representatives
Exhibit I	Point of Delivery/Interconnection Facilities
Exhibit J	Required Insurance

RENEWABLE RESOURCE CONTRACT (SCHEDULE 32)

THIS RENEWABLE RESOURCE CONTRACT (SCHEDULE 32) is entered into this 21st day of February, 2018 between Amor IX, LLC, a Delaware limited liability company (**"Seller"**) and PacifiCorp, an Oregon corporation acting in its merchant function capacity (**"PacifiCorp"**). Seller and PacifiCorp are sometimes hereinafter referred to collectively as the **"Parties"** and individually as a **"Party."**

WHEREAS, Seller intends to construct, own, operate and maintain a geothermal generation facility for the generation of electric energy located in Churchill County, Nevada, with an expected nameplate capacity rating of 20 MW (the **"Facility"**).

WHEREAS, the Parties are entering into this Agreement, as defined below, to supply energy goods and services to the University of Utah (the "**University**").

WHEREAS, Seller expects that the Facility will deliver to PacifiCorp the Net Output for supply to the University pursuant to a Renewable Energy Contract (Schedule 32) between the University and PacifiCorp (the "**University Contract**") in accordance with Utah Code Ann. § 54-17-801 and 805, and Electric Service Schedule 32.

WHEREAS, Seller desires to sell, and PacifiCorp desires to purchase, the Net Output expected to be delivered by the Facility in accordance with the terms and conditions hereof.

NOW, THEREFORE, in consideration of the foregoing and mutual promises set forth below and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties mutually agree as follows:

SECTION 1 DEFINITIONS, RULES OF INTERPRETATION

1.1 <u>Defined Terms</u>. Unless otherwise required by the context in which any term appears, initially capitalized terms used herein shall have the following meanings:

"AAA" is defined in 25.2.1.

"Abandonment" means the relinquishment of all possession and control of the Facility by Seller, other than pursuant to a transfer permitted under this Agreement, but only to the extent such relinquishment or cessation is not caused by or attributable to an Event of Default of, or request by, PacifiCorp, or an event of Force Majeure.

"Affiliate" means, with respect to any entity, each entity that directly or indirectly controls, is controlled by, or is under common control with, such designated entity, with "control" meaning the possession, directly or indirectly, of the power to direct management and policies, whether through the ownership of voting securities or by contract or otherwise.

Notwithstanding the foregoing, with respect to PacifiCorp, Affiliate shall only include Berkshire Hathaway Energy Company and its direct, wholly owned subsidiaries.

"Agreement" means this Renewable Resource Contract (Schedule 32) and any renewals hereof and any appendix, exhibit or amendment hereto.

"Availability" means, for any measurement period, the ratio, expressed as a percentage, of (a) the hours in the period during which the Facility was generating or available to generate Output divided by (b) the total number of hours in that period.

"Business Day" means any day on which banks in Salt Lake City, Utah are not authorized or required by Requirements of Law to be closed, beginning at 6:00 a.m. and ending at 5:00 p.m. local time in Utah.

"Capacity Rights" means any current or future defined characteristic, certificate, tag (but not RECs), credit, ancillary service or attribute thereof, or accounting construct, including any of the same counted towards any current or future resource adequacy or reserve requirements, associated with the electric generation capability and capacity of the Facility or the Facility's capability and ability to produce energy. Capacity Rights are measured in MW and do not include PTCs, ITCs, the Cash Grant, any Tax Credits, or any other tax incentives existing now or in the future associated with the ownership or operation of the Facility.

"Cash Grant" means the payment described in Section 1603 of the American Recovery and Reinvestment Tax Act of 2009, as such law may be amended or superseded, which payment is in lieu of receipt of ITCs.

"Commercial Operation" means when the Facility begins to operate on NV Energy's transmission system in a manner that is acceptable to NV Energy and which is expected to be normal to so operate, after energization and commissioning.

"Commission" means the Utah Public Service Commission.

"Confidential Business Information" is defined in Section 24.1.

"Contract Interest Rate" means the lesser of (a) the highest rate permitted under Requirements of Law or (b) 200 basis points per annum plus the rate per annum equal to the publicly announced prime rate or reference rate for commercial loans to large businesses in effect from time to time quoted by Citibank, N.A. as its "prime rate." If a Citibank, N.A. prime rate is not available, the applicable prime rate shall be the announced prime rate or reference rate for commercial loans in effect from time to time quoted by a bank with \$10 billion or more in assets in New York City, N.Y., selected by the Party to whom interest is being paid.

"Contract Price" means the applicable price, expressed in \$/MWh for Net Output, and Capacity Rights stated in Section 5.1.

"Contract Year" means any consecutive 12-month period during the Term, commencing at 00:00 hours on the Delivery Start Date or any of its anniversaries and ending at 24:00 hours on the last day of such 12-month period.

"Curtailment" is defined in Section 4.4.

"Default Security" is defined in Section 8.2.

"Delivery Start Date" means the date upon which Net Output is first delivered by Seller to PacifiCorp under this Agreement for delivery to University under the University Contract, after the Facility has achieved Commercial Operation and is fully interconnected, integrated, and synchronized with the transmission system of NV Energy, currently expected to be on or before September 1, 2019. Seller shall provide at least 9 months' advance written notice, if possible, to University and PacifiCorp of the anticipated Delivery Start Date, and shall promptly notify University and PacifiCorp of any expected changes to the Delivery Start Date.

"Disruption Day" means any calendar day when for any reason no Power is delivered from the Facility to PacifiCorp at the Point of Delivery during any On-Peak Hour; provided however, a Disruption Day includes only a day when Power is delivered for less than 24 hours in that calendar day. If the Facility delivers no Power to PacifiCorp at the Point of Delivery during an entire calendar day, it is not a Disruption Day.

"Effective Date" is defined in Section 2.1.

"Electric System Authority" means each of NERC, WECC, WREGIS, an RTO, a regional or sub-regional reliability council or authority, and any other similar council, corporation, organization or body of recognized standing with respect to the operations of the electric system in the WECC region.

"Emission Reduction Credit" is any credit, allowance or instrument issued or issuable pursuant to a state implementation plan under the Clean Power Plan promulgated by the Environmental Protection Agency under the Clean Air Act.

"Energy" means electric energy expressed in kilowatt hours.

"Environmental Contamination" means the introduction or presence of Hazardous Materials at such levels, quantities or location, or of such form or character, as to constitute a violation of federal, state or local laws or regulations, and present a material risk under federal, state or local laws and regulations that the Premises will not be available or usable for the purposes contemplated by this Agreement.

"EWG" means an "exempt wholesale generator," as defined under PUHCA.

"Event of Default" is defined in Section 11.1.

"Example" means an example of certain calculations to be made hereunder. Each Example is for purposes of illustration only and is not intended to constitute a representation, warranty or covenant concerning the matters assumed for purposes of each Example.

"Expected Energy" means the expected MWh of Net Output from the Facility to be delivered to the Point of Delivery as set forth in Section 6.11.1 and as specified in Exhibit A. "Facility" is defined in the Recitals and is more fully described in attached

Exhibit B.

"FERC" means the Federal Energy Regulatory Commission.

"FIN 46" is defined in Section 6.10.

"Firm Market Price Index" means (a) the average price reported on the Intercontinental Exchange ("ICE") Day-Ahead Palo Verde On-Peak Index, for on-peak hours (as specified by ICE), and (b) the average price reported on the ICE Day-Ahead Palo Verde Off-Peak Index, for off-peak hours (as determined by ICE). If either index is not available for a given period, for purposes of calculations hereunder, the Firm Market Price Index shall be deemed to equal the volumetrically-weighted average price derived from data published by ICE for the same number of days immediately preceding and immediately succeeding the period in which the index in question was not available, regardless of which days of the week are used for this purpose. If the Firm Market Price Index or its replacement or any component of that index or its replacement ceases to be published or available, or useful for its intended purpose hereunder, during the Term, the Parties shall agree upon a replacement Firm Market Price Index or component an index or component that, after any necessary adjustments, provides the most reasonable substitute quotation of the daily price of electricity for the applicable periods.

"Force Majeure" is defined in Section 15.1.

"Forced Outage" means NERC Event Types U1, U2 and U3, as set forth in attached Exhibit C, and specifically excludes any Maintenance Outage or Planned Outage.

"Generation Interconnection Agreement" means the large generator interconnection agreement to be entered into separately between Seller and Interconnection Provider concerning the Interconnection Facilities.

"Governmental Authority" means any supranational, federal, state or other political subdivision thereof, having jurisdiction over Seller, PacifiCorp or this Agreement, including any municipality, township or county, and any entity or body exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including any corporation or other entity owned or controlled by any of the foregoing.

"Guaranteed Energy" has the meaning set forth in 6.11.1.

"Guaranteed Power" has the meaning set forth in 6.11.2

"Hazardous Materials" means any waste or other substance that is listed, defined, designated or classified as or determined to be hazardous under or pursuant to any environmental law or regulation.

"Indemnified Party" is defined in Section 6.1.3(b).

"Indemnifying Party" is defined in Section 6.1.3(b).

"Interconnection Facilities" means all the facilities installed, or to be installed, for the purpose of interconnecting the Facility to the System, including electrical transmission lines, upgrades, transformers and associated equipment, substations, relay and switching equipment, and safety equipment.

"Interconnection Provider" means NV Energy.

"ITCs" means the investment tax credits established pursuant to Section 48 of the Internal Revenue Code, as such law may be amended or superseded.

"Lender" means an entity lending money or extending credit (including any financing lease, monetization of tax benefits, transaction with a tax equity investor, back leverage financing or credit derivative arrangement) to Seller or Seller's Affiliates, for purposes relating to the Facility, which shall include, without limitation, (a) for the construction, term or permanent financing or refinancing of the Facility; (b) for working capital or other ordinary business requirements for the Facility (including for the maintenance, repair, replacement or improvement of the Facility); (c) for any development financing, bridge financing, credit support, and related credit enhancement or interest rate, currency, weather, or Renewable and Environmental Attributes in connection with the development, construction or operation of the Facility; or (d) for the purchase of the Facility and related rights from Seller.

"Letter of Credit" means, in the case of Default Security, an irrevocable standby letter of credit in a form and from an issuer reasonably acceptable to PacifiCorp, naming PacifiCorp as the party entitled to demand payment and present draw requests thereunder, which by its terms will remain in effect for at least 90 days after the end of the specified period for which the letter of credit is to be maintained.

"Liabilities" is defined in Section 13.1.1.

"Liquidated Damages" means the amounts specified in 6.11.1 or 6.11.2, as applicable.

"Maintenance Outage" means NERC Event Type MO, as set forth in attached Exhibit C, and includes any outage that is not a Forced Outage or a Planned Outage.

"Maximum Delivery Rate" means the maximum hourly rate of delivery of Net Output in MWh from the Facility to the Point of Delivery, calculated on the basis of the Net Output delivered in an hour accruing at an average rate equivalent to the actual Nameplate Capacity Rating.

"Mediation Notice" is defined in Section 25.2.1(a).

"Mediation Procedures" is defined in Section25.2.1.

"Moody's" means Moody's Investor Services, Inc.

"MW" means megawatt.

"MWh" means megawatt hour.

"Nameplate Capacity Rating" means the maximum installed instantaneous generation capacity of the completed Facility, expressed in MW, when operated in compliance with the Generation Interconnection Agreement and consistent with manufacturer's recommended operating parameters. The Nameplate Capacity Rating of the Facility is 20 MW, net of station use, transformation and transmission losses to the Point of Delivery.

"NERC" means the North American Electric Reliability Corporation.

"Net Output" means the Output actually delivered to the Point of Delivery by Seller through NV Energy for sale to PacifiCorp and resale to University, rounded to whole MW increments. Net Output for each hour shall be measured as the amount of Scheduled Delivery for that hour.

"Network Resource" is defined in the Network Service Provider's Tariff.

"Network Service Provider" means PacifiCorp, acting through its transmission service function.

"On-Peak Hours" means those hours defined as On-Peak in PacifiCorp's Utah Schedule 32. At the time of execution of this Agreement, On-Peak hours are 7 a.m. to 11 p.m., inclusive, Monday through Friday, except holidays from October through April, inclusive; and 1 p.m. to 9 p.m., inclusive, Monday through Friday, except holidays, from May through September, inclusive. On-Peak Hours are based on the prevailing Mountain Time, including the effects of Daylight Savings Time.

"Output" means all Power and Energy produced by the Facility.

"PacifiCorp" is defined in the Recitals.

"PacifiCorp Indemnitees" is defined in Section 13.1.1.

"PacifiCorp Representatives" is defined in Section 6.12.

"PacifiCorp's Cost to Cover" means the positive difference, if any, between (a) the sum of (i) the time weighted average of the Firm Market Price Index for each day for which the determination is being made, minus (b) the Contract Price specified in Exhibit D in effect on such days, stated as an amount per MWh. If on a given day the difference between (a) minus (b) referenced above is zero or negative, then PacifiCorp's Cost to Cover shall be zero dollars (\$0), and Seller shall have no obligation to pay any amount to PacifiCorp on account of Section 11.2.1 with respect to such day.

"Party" and "Parties" are defined in the Recitals.

"Permits" means the permits, licenses, approvals, certificates, entitlements and other authorizations issued by Governmental Authorities required for the construction, ownership and operation of the Facility and occupancy of the Premises specified in Exhibit E, and all amendments, modifications, supplements, general conditions and addenda thereto.

"Planned Outage" means NERC Event Type PO, as set forth in attached Exhibit C, and specifically excludes any Maintenance Outage or Forced Outage.

"Point of Delivery" means the Gon.PAV where Seller shall deliver Net Output to PacifiCorp utilizing firm Point-to-Point Transmission Services provided by Transmission Provider.

"Point of Interconnection" means the Ragtown Substation on the NV Energy transmission system.

"Portfolio Energy Credit" or "PEC" means the portfolio energy credit certified pursuant to Nevada Revised Statutes section 704.7803 from energy required to operate the Soda Lake geothermal project consistent with the Declaratory Order of the Public Utilities Commission of Nevada in Docket No. 12-09015, dated November 12, 2012.

"Power" means electric power expressed in kilowatts.

"Premises" means the real property on which the Facility is or will be located, as more fully described on Exhibit B.

"Prudent Electrical Practices" means any of the practices, methods and acts engaged in or approved by a significant portion of the independent electric power generation industry for geothermal facilities of similar size and characteristics or any of the practices, methods or acts, which, in the exercise of reasonable judgment in the light of the facts known at the time a decision is made, could have been expected to accomplish the desired result at the lowest reasonable cost consistent with reliability, safety and expedition. Prudent Electrical Practices is not intended to be limited to the optimum practice, method or act to the exclusion of all others, but rather to be a spectrum of possible practices, methods or acts.

"PTCs" means production tax credits under Section 45 of the Internal Revenue Code, as such law may be amended or superseded.

"PUHCA" means the Public Utility Holding Company Act of 2005.

"PURPA" means the Public Utility Regulatory Policies Act of 1978.

"Qualifying Institution" means a United States commercial bank or trust company organized under the laws of the United States of America or a political subdivision thereof, having assets of at least \$10,000,000,000 (net of reserves) and a credit rating on its long-term senior unsecured debt of at least "A" from S&P and "A2" from Moody's.

"RECs" means (a) all Renewable and Environmental Attributes associated with all Output, together with (b) REC Reporting Rights associated with such Output and Renewable and Environmental Attributes, however commercially transferred or traded under any or other product names, such as "Renewable Energy Credits," "Green-e Certified," "Green Tags" or otherwise. One REC represents the Renewable and Environmental Attributes made available by the generation of one MWh of energy from the Facility, excluding PECs associated with station service.

"REC Reporting Rights" means the exclusive right of a purchaser of Renewable and Environmental Attributes to report ownership of Renewable and Environmental Attributes in compliance with federal or state law, if applicable, and to federal or state agencies or other parties at such purchaser's discretion, and include reporting under Section 1605(b) of the Energy Policy Act of 1992, or under any present or future domestic, international, or foreign emissions trading program or renewable portfolio standard.

"Renewable and Environmental Attributes" means any and all claims, credits, benefits, emissions reductions, offsets, and allowances, howsoever entitled, resulting from the avoidance of the emission of any gas, chemical, or other substance to the air, soil or water. Renewable and Environmental Attributes include but are not limited to any and all: (a) avoided emissions of pollutants to the air, soil, or water such as (subject to the foregoing) sulfur oxides (SOx), nitrogen oxides (NOx), carbon monoxide (CO), and other pollutants; (b) Emission Reduction Credits; and (c) avoided emissions of carbon dioxide (CO2), methane (CH4), and other greenhouse gases (GHGs) that have been determined by the United Nations Intergovernmental Panel on Climate Change to contribute to the actual or potential threat of altering the Earth's climate by trapping heat in the atmosphere. Renewable and Environmental Attributes do not include (i) PTCs, ITCs or any Tax Credits, or certain other tax incentives existing now or in the future associated with the construction, ownership or operation of the Facility, (ii) adverse wildlife or environmental impacts, or (iii) Portfolio Energy Credits.

"Renewable Energy Supply Agreement" or "RESA" means the Renewable Energy Supply Agreement of even date herewith between University and Seller.

"Reporting Month" is defined in Section 6.9.1.

"Required Facility Documents" means the Permits and other authorizations, rights and agreements now or hereafter necessary for construction, operation, and maintenance of the Facility set forth in Exhibit F. Nothing set forth in Exhibit F limits the obligations of Seller to obtain the Permits set forth in Exhibit E or otherwise required hereunder.

"Requirements of Law" means any applicable and mandatory (but not merely advisory) federal, state and local law, statute, regulation, rule, action, order, code or ordinance enacted, adopted, issued or promulgated by any federal, state, local or other Governmental Authority or regulatory body (including those pertaining to electrical, building, zoning, environmental and occupational safety and health requirements).

"RTO" means any entity that becomes responsible as system operator for, or directs the operation of, the System.

"S&P" means Standard & Poor's Rating Group (a division of McGraw-Hill, Inc.).

"Scheduled Delivery" for each hour means the MW of Power (and associated Energy) scheduled by Seller from the Facility and delivered to PacifiCorp for redelivery to University through NV Energy at the Point of Delivery, in whole MW increments, in accordance with all applicable scheduling guidelines and requirements of the North American Electric Reliability Corporation (NERC) and NV Energy.

"Seller" is defined in the Recitals.

"Seller Indemnitees" is defined in Section 13.1.2.

"Seller's Cost to Cover" means the positive difference, if any, between (a) the Contract Price per MWh specified in Exhibit D, and (b) the net proceeds per MWh actually realized by Seller from the sale to a third party of Net Output not purchased by PacifiCorp or RECs not sold to University as required hereunder. If on any given day the difference between (a) minus (b) referenced above is zero or negative, then Seller's Cost to Cover shall be zero dollars with respect to such day, and PacifiCorp shall have no obligation to pay any amount to Seller on account of Section 11.2.2.

"Seller Excused Delivery Hour" means any hour during a calendar day in which the Facility was unable to deliver Net Output to PacifiCorp (or during which PacifiCorp was unable to accept such delivery) due to one or more of the following events, to the extent not caused by Seller's actions or negligence, each as recorded by Seller's SCADA and indicated by Seller's electronic fault log: (a) an Event of Force Majeure; (b) Planned Outages, as defined in this Agreement, in up to a maximum of three calendar days in any Contract Year, other than every sixth Contract Year after the Delivery Start Date, during which Planned Outages may include up to seven calendar days for Seller to overhaul its turbine, consistent with the applicable operating manual; (c) unplanned outages or downtime of the Facility in up to a maximum of three calendar days per Contract Year; (d) other conditions that prevent delivery by Seller or acceptance by PacifiCorp of Net Output in up to a maximum of seven calendar days per Contract Year, or (e) a default by PacifiCorp under the Renewable Energy Contract or this Agreement that prevents the delivery of power from Seller; provided, however, that if any of the events described above in items (a) through (e) occur simultaneously, then the relevant hour shall only be counted once in order to prevent double counting. A Seller Excused Delivery Hour shall not include any hour when (i) the Facility or any portion thereof was unavailable due to Seller's nonconformance with the applicable generation interconnection agreement or (ii) the Facility or any portion thereof was paused or withdrawn from use by Seller for reasons other than those covered in this definition. All hours qualifying for each category of Seller Excused Delivery Hour as specified above shall be summed, up to the applicable maximum, if any, specified above. All Seller Excused Delivery Hours will be considered for purposes of determining Guaranteed Energy liquidated damages under Section 6.11.1. Only those Seller Excused Delivery Hours that occur during an On-Peak Hour will be considered for purposes of calculating Guaranteed Power liquidated damages under Section 6.11.2. Any calendar day in which Seller delivers no Power to PacifiCorp at the Point of Delivery will not be counted against the calendar days set forth in subsections (c) unplanned outages or (d) other conditions, set forth above.

"System" means the electric transmission substation and transmission or distribution facilities owned, operated or maintained by Transmission Provider, which include

the circuit reinforcements, extensions, and associated terminal facility reinforcements or additions required to interconnect the Facility, all as set forth in the Generation Interconnection Agreement.

"Tariff" means the PacifiCorp FERC Electric Tariff Fifth Revised Volume No. 11 Pro Forma Open Access Transmission Tariff, as revised from time to time.

"Tax Credits" means any state, local and/or federal production tax credit, tax deduction, and/or investment tax credit specific to the production of renewable energy and/or investments in renewable energy facilities.

"Technical Dispute Notice" is defined in Section 25.2.2(a).

"Technical Dispute Procedures" is defined in Section 25.2.2.

"Technical Expert" is defined in Section 25.2.2.

"Term" is defined in Section 2.1.

"Transmission Provider" means NV Energy.

"Transmission Service" means the transmission services pursuant to which the Transmission Provider transmits Output to the Point of Delivery, as applicable.

"University" is defined in the Recitals.

"University Contract" is defined in the Recitals.

"University Damages" means either (i) Liquidated Damages as defined in a RESA, as applicable, or (ii) an amount equal to the present value of the economic loss to University, if any, resulting from an Event of Default by Seller under this Agreement for so long as such Event of Default continues, or for the entire remaining Term in the event of termination, determined in a commercially reasonable manner and without duplication of any costs included in calculating PacifiCorp's Cost to Cover. Factors that may be used in determining University Damages may include, without limitation, the incremental cost if PacifiCorp is able to obtain replacement Power and Energy from a comparable baseload renewable resource reasonably acceptable to the University that can be delivered to the University under the University Contract, and incremental daily fees or costs owed by University to PacifiCorp under the University Contract, all of which shall be calculated for the applicable period, and shall specifically include and reflect the value of comparable renewable resources, including associated Renewable and Environmental Attributes.

"WECC" means the Western Electricity Coordinating Council.

"WREGIS" means the Western Renewable Energy Generation Information

System.

"WREGIS Certificate" means "Certificate" as defined by WREGIS in the WREGIS Operating Rules.

"WREGIS Operating Rules" means the operating rules and requirements adopted by WREGIS.

1.2 <u>Rules of Interpretation</u>.

1.2.1 <u>General</u>. Unless otherwise required by the context in which any term appears or otherwise stated herein, (a) the singular includes the plural and vice versa; (b) references to "Articles," "Sections," "Schedules," "Annexes," "Appendices" or "Exhibits" are to articles, sections, schedules, annexes, appendices or exhibits hereof; (c) all references to a particular entity or an electricity market price index include a reference to such entity's or index's successors; (d) "herein," "hereof" and "hereunder" refer to this Agreement as a whole; (e) all accounting terms not specifically defined herein shall be construed in accordance with generally accepted accounting principles, consistently applied; (f) the masculine includes the feminine and neuter and vice versa; (g) "including" means "including, without limitation" or "including, but not limited to"; (h) all references to a particular law or statute mean that law or statute as amended from time to time; (i) the word "or" is not necessarily exclusive; and (j) references to "days" are to calendar days.

1.2.2 <u>Terms Not to be Construed For or Against Either Party</u>. Each term hereof shall be construed simply according to its fair meaning and not strictly for or against either Party. The Parties have jointly prepared this Agreement, and no term hereof shall be construed against a Party on the ground that the Party is the author of that provision.

1.2.3 <u>Headings</u>. The headings used for the sections and articles hereof are for convenience and reference purposes only and shall in no way affect the meaning or interpretation of the provisions hereof.

1.2.4 <u>Examples</u>. Example calculations and other examples set forth herein are for purposes of illustration only and are not intended to constitute a representation, warranty or covenant concerning the example itself or the matters assumed for purposes of such example. If there is a conflict between an example and the text hereof, the text shall control.

SECTION 2 TERM

2.1 <u>Term</u>. This Agreement shall become effective the date upon which (a) this Agreement has been fully executed and delivered by both Parties (b) final and non-appealable Commission orders have been entered approving without material modification all of the terms of this Agreement and the University Contract and (c) the Project Development Security as set forth in Section 5.1 of the RESA has been posted (the "Effective Date"), and, unless earlier terminated as provided herein,

(the "Term"), subject to extension of this Agreement upon mutual agreement of the Parties.

2.2 <u>Project</u>.

(a) Seller hereby represents and warrants to PacifiCorp that the Required Facility Documents including, but not limited to the material permits, consents and agreements necessary to operate and maintain the Facility will be obtained by Seller prior to Commercial Operation.

(b) Seller shall not undertake to increase the Output of the Facility without the prior written consent of PacifiCorp.

2.3 <u>PacifiCorp's Right to Monitor</u>. During the Term, Seller shall permit PacifiCorp and its advisors and consultants to:

(a) Review and discuss with Seller and its advisors and consultants concerning operation of the Facility.

(b) Monitor the operation of the Facility and the performance of the Seller's contractors.

(c) Nothing in this Agreement shall be construed to require PacifiCorp to review, comment on, or approve of any contract between Seller and a third party.

(d) Perform such examinations, inspections, and quality surveillance as, in PacifiCorp's reasonable judgment, are appropriate and advisable.

With respect to PacifiCorp's right to monitor under this Section 2.3, (i) PacifiCorp is under no obligation to exercise any of these monitoring rights, (ii) such monitoring shall occur subject to reasonable rules developed by Seller regarding Facility access, health, safety, and environmental requirements, and (iii) PacifiCorp shall have no liability to Seller for failing to advise it of any condition, damages, circumstances, infraction, fact, act, omission or disclosure discovered or not discovered by PacifiCorp with respect to the Facility or any contractor. PacifiCorp shall maintain one or more designated representatives for purposes of the monitoring activities contemplated in this Section 2.3, which representatives shall have authority to act for PacifiCorp in all technical matters under this Section 2.3 as authorized by PacifiCorp but not to amend or modify any provision hereof. PacifiCorp's initial representatives and their contact information are listed in Exhibit H. PacifiCorp may, by written notice to Seller, change its representatives or their contact information.

2.4 <u>Tax Credits</u>. Seller shall bear all risks, financial and otherwise throughout the Term, associated with Seller's or the Facility's eligibility to receive PTCs, ITCs, the Cash Grant, or other Tax Credits, or to qualify for accelerated depreciation for Seller's accounting, reporting or tax purposes. The obligations of the Parties hereunder, including those obligations set forth herein regarding the purchase and price for and Seller's obligation to deliver Net Output, shall be effective regardless of whether the sale of Output or Net Output from the Facility is eligible for, or receives, PTCs, ITCs, the Cash Grant, or other Tax Credits during the Term.

SECTION 3 REPRESENTATIONS AND WARRANTIES

3.1 <u>Mutual Representations and Warranties</u>. Each Party represents, covenants, and warrants to the other that:

3.1.1 <u>Organization</u>. It is duly organized and validly existing under the laws of the State of its organization.

3.1.2 <u>Authority</u>. It has the requisite power and authority to enter hereinto and to perform according to the terms hereof.

3.1.3 <u>Corporate Actions</u>. It has taken all corporate actions required to be taken by it to authorize the execution, delivery and performance hereof and the consummation of the transactions contemplated hereby.

3.1.4 <u>No Contravention</u>. The execution and delivery hereof do not contravene any provision of, or constitute a default under, any indenture, mortgage, security instrument or undertaking, or other material agreement to which it is a party or by which it is bound, or any valid order of any court, or any regulatory agency or other body having authority to which it is subject.

3.1.5 <u>Valid and Enforceable Agreement</u>. This Agreement is a valid and legally binding obligation of it, enforceable against it in accordance with its terms, except as the enforceability hereof may be limited by general principles of equity or bankruptcy, insolvency, bank moratorium or similar laws affecting creditors' rights generally and laws restricting the availability of equitable remedies.

3.1.6 <u>Eligible Contract Participant</u>. It, and any guarantor of its obligations hereunder, is an "eligible contract participant" within the meaning of the Commodity Exchange Act.

3.2 <u>Seller's Further Representations and Warranties</u>. Seller further represents, covenants, and warrants to PacifiCorp that:

3.2.1 <u>Authority</u>. Seller (a) has or will have 30 days prior to Commercial Operation all required regulatory authority to make wholesale sales from the Facility; (b) has the power and authority to own and operate the Facility and be present upon the Premises for the Term; and (c) is duly qualified and in good standing under the laws of each jurisdiction where its ownership, lease or operation of property or the conduct of its business requires such qualification.

3.2.2 <u>No Contravention</u>. The execution, delivery, performance and observance by Seller of its obligations hereunder do not and will not:

(a) contravene, conflict with or violate any provision of any material Requirements of Law presently in effect having applicability to either Seller or any of Seller's members;

(b) require the consent or approval of or material filing or registration with any Governmental Authority or other person other than such consents and approvals which are (i) set forth in Exhibit K or (ii) required in connection with the operation of the Facility and expected to be obtained in due course;

(c) result in a breach of or constitute a default under any provision of any security issued by any of Seller's members or managers, the effect of which would materially and adversely affect Seller's performance of, or ability to perform, its obligations hereunder, or any material agreement, instrument or undertaking to which either Seller's members or any Affiliates of Seller's members is a party or by which the property of any of Seller's members or any Affiliates of Seller's members is bound, the effect of which would materially and adversely affect Seller's performance of, or ability to perform, its obligations hereunder.

3.2.3 <u>Litigation</u>. No litigation, arbitration, investigation or other proceeding is pending or, to the best of Seller's knowledge, threatened in writing against any of Seller or its members, with respect hereto and the transactions contemplated hereunder. No other investigation or proceeding is pending or threatened in writing against Seller, its members, or any Affiliate, the effect of which would materially and adversely affect Seller's performance of its obligations hereunder.

3.2.4 <u>Required Facility Documents</u>. All Required Facility Documents are listed on Exhibit F. Seller shall hold the Required Facility Documents as of Commercial Operation, and shall maintain for the Term all material rights and entitlements necessary to own and operate the Facility and to deliver Net Output to PacifiCorp in accordance with this Agreement. The anticipated use of the Facility complies with all applicable restrictive covenants affecting the Premises. Seller shall notify PacifiCorp of any additional material consent or approval that is required for the operation and maintenance of the Facility promptly after Seller makes any such determination.

3.2.5 <u>Delivery of Energy</u>. Seller shall hold firm point-to-point Transmission Service rights and other rights sufficient to enable Seller to deliver Net Output based on the Nameplate Capacity Rating from the Facility to the Point of Delivery pursuant to this Agreement beginning upon Commercial Operation and throughout the Term. Provided Seller has tendered to PacifiCorp information required by Section 29.2 of the Tariff, PacifiCorp will request, within five (5) calendar days of execution of this Agreement, designation of this Agreement as a Network Resource at the Point of Delivery under PacifiCorp's Network Integration Transmission Service Agreement with the Network Service Provider. The Network Service Provider processes Network Resource designation requests in accordance with the Tariff.

3.2.6 <u>Transmission and Interconnection Costs</u>. Section I.C.5 of Rocky Mountain Power Electric Service Schedule No. 32 requires that Seller will be responsible for all interconnection and transmission integration costs. This means Seller must be responsible for any network upgrade costs or other costs associated with: (i) Seller's generator interconnection; (ii) Seller's energy delivery obligations under Section 3.2.5 of this Agreement; and (iii) PacifiCorp's request to designate this Agreement as a Network Resource. The Contract Price assumes that no network upgrades will be required. Thus, unless the Parties otherwise agree, and notwithstanding anything to the contrary in this Agreement, this Agreement will terminate unless prior to the Effective Date: (i) Network Service Provider confirms through the Tariff study process that no network upgrades on the Network Service Provider's transmission system will be required for Seller's generator interconnection or for PacifiCorp's request to designate this Agreement as a Network Resource.

3.2.7 <u>Permits</u>. All Permits and all other material permits, consents, approvals, licenses and authorizations required for Seller's performance of this Agreement shall be obtained prior to Commercial Operation and remain in full force and effect. Seller shall notify PacifiCorp of any additional material permit, consent, approval, license or authorization that is required for Seller's performance of this Agreement promptly after Seller makes any such determination.

3.2.8 <u>Leases</u>. All leases of real property required for the operation of the Facility or the performance of any obligations of Seller hereunder are set forth and accurately described in Exhibit G. Upon request by PacifiCorp, Seller shall provide copies of the leases to PacifiCorp which shall be Confidential Business Information. Seller's leases, licenses or other grants of rights in real property required for the operation of the Facility or location of Facility on the Premises shall be valid through the Term and Seller is not in breach of any terms or conditions of such leases or other rights in real property for the Facility or Premises.

3.2.9 <u>Green Guides</u>. Seller has and will at all times be fully compliant with the requirements of the Federal Trade Commission's "Green Guides," 77 F.R. 62122, 16 C.F.R. Part 260, as amended or restated in any communication concerning Net Output, the Facility or the RECs.

3.2.10 <u>Undertaking of Agreement; Professionals and Experts</u>. Seller has engaged those professional or other experts it believes necessary to understand its rights and obligations pursuant to this Agreement. All professionals or experts including, but not limited to, engineers, attorneys or accountants, that Seller may have consulted or relied on in undertaking the transactions contemplated by this Agreement have been solely those of Seller. In entering into this Agreement and the undertaking by Seller of the obligations set forth herein, Seller has investigated and determined that it is capable of performing hereunder and has not relied upon the advice, experience or expertise of PacifiCorp in connection with the transactions contemplated by this Agreement.

3.2.11 <u>Verification</u>. All information relating to the Facility, its operation and Output and the Premises provided to PacifiCorp and contained in this Agreement has been verified by Seller and is true and accurate.

3.3 <u>PacifiCorp's Further Representations and Warranties</u>. PacifiCorp further represents, covenants, and warrants to Seller that:

3.3.1 <u>Authority</u>. PacifiCorp has all required regulatory authority to purchase the Net Output sold to it by Seller hereunder.

3.3.2 <u>No Contravention</u>. The execution, delivery, performance and observance by PacifiCorp of its obligations hereunder do not and will not require the consent or

approval of any Governmental Authority other than such consents and approvals which are expected to be obtained in due course.

3.3.3 <u>Litigation</u>. No litigation, arbitration, investigation or other proceeding is pending or, to the best of PacifiCorp's knowledge, threatened against PacifiCorp or its Affiliates with respect hereto and the transactions contemplated hereunder, the effect of which would materially and adversely affect PacifiCorp's performance of its obligations hereunder.

3.4 <u>No Other Representations or Warranties</u>. Each Party acknowledges that it has entered hereinto in reliance upon only the representations and warranties set forth in this Agreement, and that no other representations or warranties have been made by the other Party with respect to the subject matter hereof.

3.5 <u>Continuing Nature of Representations and Warranties; Notice</u>. The representations and warranties set forth in this Section 3 are made as of the Effective Date. If at any time during the Term, any Party obtains actual knowledge of any event or information which causes any of the representations and warranties in Section 3 to be materially untrue or misleading, such Party shall provide the other Party with written notice of the event or information, the representations and warranties affected, and the action, if any, which such Party intends to take to make the representations and warranties true and correct. The notice required pursuant to this Section 3 shall be given as soon as practicable after the occurrence of each such event.

SECTION 4 DELIVERIES OF NET OUTPUT

4.1 <u>Purchase and Sale</u>. Except as otherwise expressly provided herein, commencing upon Commercial Operation and continuing through the Term, Seller shall (a) sell and make available to PacifiCorp, and PacifiCorp shall purchase and receive the entire Net Output from the Facility at the Point of Delivery as measured pursuant to Scheduled Delivery, to enable PacifiCorp to sell and deliver Net Output to University under the University Contract; and (b) sell and transfer to University, at no additional cost to PacifiCorp or University, all Renewable and Environmental Attributes and RECs associated with the Output. PacifiCorp shall be under no obligation to make any purchase hereunder other than Net Output as described above. PacifiCorp shall not be obligated to purchase, receive or pay for Output that is not delivered to the Point of Delivery.

4.2 <u>No Sales to Third Parties</u>. During the Term, Seller shall not sell any Output, RECs or Capacity Rights from the Facility to any party other than PacifiCorp or University; provided, however, that this restriction shall not apply during periods when PacifiCorp is in default hereof because it has failed to accept or purchase Net Output as required to hereunder. Seller shall be able to sell Net Output and RECs to others during Curtailments. In event Seller sells Net Output or RECs to third parties during Curtailments under this Section 4.2, Seller shall be responsible for procurement of any services, including transmission service, to complete such sales to any third parties. 4.3 <u>Title and Risk of Loss of Net Output</u>. Seller shall deliver Net Output to the Point of Delivery free and clear of all liens, claims and encumbrances. Title to and risk of loss of all Net Output shall transfer from Seller to PacifiCorp upon its delivery to PacifiCorp at the Point of Delivery. Seller shall be deemed to be in exclusive control of, and responsible for, any damage or injury caused by, all Output up to and at the Point of Delivery. PacifiCorp shall be deemed to be in exclusive control of, and responsible for, any damages or injury caused by, Net Output from and after the Point of Delivery.

4.4 Curtailment. PacifiCorp shall not be obligated to purchase, receive, pay for, or pay any damages associated with, Net Output (or associated Production Tax Credits) if such Net Output is not delivered to the Point of Delivery for any reason ("Curtailment"), including but not limited to any of the following: (a) the interconnection between the Facility and the System is disconnected, suspended or interrupted, in whole or in part, under the terms of the Generation Interconnection Agreement or otherwise, (b) the Transmission Provider or Network Service Provider directs a general curtailment, reduction, or redispatch of generation in the area, (which would include the Net Output) for any reason, even if such curtailment or redispatch directive is carried out by PacifiCorp as Network Service Provider, which may fulfill such directive by acting in its sole discretion; or if PacifiCorp curtails or otherwise reduces the Net Output in order to meet its obligations to the Transmission Provider or Network Service Provider to operate within system limitations, (c) the Facility's Output is not received because the Facility is not fully integrated or synchronized with the System, or (d) an event of Force Majeure prevents either Party from delivering or receiving Net Output. However, PacifiCorp shall be obligated to purchase, receive, pay for or pay any damages associated with Net Output if Seller obtains alternative transmission rights during a Curtailment to another location on the PacifiCorp (PACE) system at which PacifiCorp can purchase and use an alternative transmission arrangement to deliver Net Output to University at no incremental cost to PacifiCorp or University. PacifiCorp's ability to deliver Net Output to University using alternative transmission arrangements may be subject to the results of a request for a transmission service modification and transmission service study, if applicable. Seller must have the rights to available transmission capacity on the Transmission Provider and the Network Service Provider systems.

4.5 <u>PacifiCorp as Merchant</u>. Seller acknowledges that PacifiCorp, acting in its merchant capacity function as purchaser under this Agreement, has no responsibility for or control over Transmission Provider or Network Service Provider.

4.6 <u>Title to RECs</u>. All RECs and Emission Reduction Credits associated with the Output will be deposited by Seller into an account maintained by or on behalf of, and will be retired or otherwise dealt with as directed by, University. PacifiCorp and Seller waive any claim to ownership of any RECs or Emission Reduction Credits under this Agreement. University shall have all REC Reporting Rights.

4.7 Capacity Rights.

4.7.1 <u>Purchase and Sale of Capacity Rights</u>. For and in consideration of PacifiCorp's agreement to purchase from Seller the Facility's Net Output on the terms and conditions set forth herein, Seller transfers to PacifiCorp, and PacifiCorp accepts from Seller,

any right, title, and interest that Seller may have in and to Capacity Rights, if any, existing during the Term.

4.7.2 <u>Representation Regarding Ownership of Capacity Rights</u>. Seller represents that it has not sold, and covenants that during the Term it will not sell or attempt to sell to any other person or entity the Capacity Rights, if any. During the Term, Seller shall not report to any person or entity that the Capacity Rights, if any, belong to anyone other than PacifiCorp. PacifiCorp may at its own risk and expense report to any person or entity that Capacity Rights exclusively belong to it.

4.7.3 <u>Further Assurances</u>. At PacifiCorp's request, the Parties shall execute such documents and instruments as may be reasonably required to affect recognition and transfer of the Capacity Rights, if any, to PacifiCorp.

4.8 <u>Third Party Sales of PECs</u>. Seller may sell PECs to any qualified buyer provided that such PECs are associated with energy required to operate the Facility, including station usage and parasitic load.

SECTION 5 CONTRACT PRICE; COSTS

5.1 <u>Contract Price; Includes Capacity Rights.</u> PacifiCorp shall pay Seller the prices stated below for all deliveries of Net Output and associated RECs, up to the Maximum Delivery Rate. The Contract Price provided for in this Section 5.1 includes the consideration to be paid to Seller for all Net Output, Capacity Rights and RECs, and Seller shall not be entitled to any compensation over and above the Contract Price, as the case may be, for Capacity Rights or RECs associated therewith. For the period beginning on the Delivery Start Date and thereafter during the Term, PacifiCorp shall pay to Seller the Contract Price per MWh of Net Output delivered to the Point of Delivery, as specified in Exhibit D. During testing prior to Commercial Operation, Seller shall have the right to sell and deliver Net Output to NV Energy as test energy.

5.2 <u>Costs and Charges</u>. Seller shall be responsible for paying or satisfying when due all costs or charges imposed in connection with the scheduling and delivery of Output up to and at the Point of Delivery, including transmission costs, Transmission Service, and transmission line losses, and any operation and maintenance charges imposed by Interconnection Provider and Transmission Provider for the Interconnection Facilities. PacifiCorp shall be responsible for all costs or charges, if any, imposed in connection with the delivery of Net Output from and after the Point of Delivery, including transmission costs and transmission line losses and imbalance charges or penalties. Without limiting the generality of the foregoing, Seller, in accordance with the Generation Interconnection Agreement, shall bear all costs associated with the modifications to Interconnection Facilities or the System (including system upgrades) caused by or related to (a) the interconnection of the Facility with the System and (b) any increase in generating capacity of the Facility.

5.3 <u>Station Service</u>. Seller shall be responsible for arranging and obtaining, at its sole risk and expense, any station service required by the Facility that is not provided by the Facility itself.
5.4 <u>Taxes</u>. Seller shall pay or cause to be paid when due, or reimburse PacifiCorp for, all existing and any new sales, use, excise, severance, ad valorem, and any other similar taxes, imposed or levied by any Governmental Authority up to and including, but not beyond, the Point of Delivery, on the generation of Net Output, Capacity Rights or on the sale of Net Output, Capacity Rights from Seller to PacifiCorp hereunder, regardless of whether such taxes are imposed on PacifiCorp or Seller under Requirements of Law. PacifiCorp shall pay or cause to be paid when due all such taxes are imposed on PacifiCorp or Seller under Requirements of Delivery upon a purchaser of power, regardless of whether such taxes are imposed on PacifiCorp or Seller under Requirements of Law. The Contract Price shall not be adjusted on the basis of any action of any Governmental Authority with respect to changes to or revocations of sales and use tax benefits, rebates, exception or give back. In the event any taxes are imposed on a Party for which the other Party is responsible hereunder, the Party on which the taxes are imposed shall promptly provide the other Party notice thereof and such other information as such Party may reasonably request with respect to any such taxes.

5.5 <u>Costs of Ownership and Operation</u>. Without limiting the generality of any other provision hereof and subject to Section 5.4, Seller shall be solely responsible for paying when due (a) all costs of owning and operating the Facility in compliance with existing and future Requirements of Law and the terms and conditions hereof, and (b) all taxes and charges (however characterized) now existing or hereinafter imposed on or with respect to the Facility, its operation, or on or with respect to emissions or other environmental impacts of the Facility, including any such tax or charge (however characterized) to the extent payable by a generator of such energy or Renewable and Environmental Attributes.

5.6 <u>Rates Not Subject to Review</u>. The rates for service specified herein shall remain in effect until expiration of the Term, and shall not be subject to change for any reason, including regulatory review, absent agreement of the parties. Neither Party shall petition FERC pursuant to the provisions of Sections 205 or 206 of the Federal Power Act (16 U.S.C. § 792 et seq.) to amend such prices or terms, or support a petition by any other person seeking to amend such prices or terms, absent the agreement in writing of the other Party. Further, absent the agreement in writing by both Parties, the standard of review for changes hereto proposed by a Party, a nonparty or the FERC acting sua sponte shall be the "public interest" application of the "just and reasonable" standard of review set forth in <u>United Gas Pipe Line Co. v. Mobile Gas Service Corp.</u>, 350 U.S. 332 (1956) and <u>Federal Power Commission v. Sierra Pacific Power Co.</u>, 350 U.S. 348 (1956) and clarified by <u>Morgan Stanley Capital Group</u>. Inc. v. Public Util. Dist. No. 1 of Snohomish, 554 U.S. 527, 128 S. Ct. 2733 (2008).

SECTION 6 OPERATION AND CONTROL

6.1 <u>Standard of Facility Operation</u>.

6.1.1 <u>General</u>. At Seller's sole cost and expense, Seller shall build, operate, maintain and repair the Facility and the Interconnection Facilities owned by Seller in accordance with (a) the applicable and mandatory standards, criteria and formal guidelines of FERC, NERC, any RTO, and any other Electric System Authority and any successors to the functions thereof; (b) the Permits and Required Facility Documents; (c) the Generation Interconnection Agreement;

(d) all Requirements of Law; (e) the requirements hereof; and (f) Prudent Electrical Practice. Seller acknowledges that it shall have no claims hereunder against PacifiCorp hereunder with respect to any requirements imposed by or damages caused by (or allegedly caused by) the Transmission Provider. Seller will have no claims against PacifiCorp under this Agreement with respect to the provision of station service.

6.1.2 <u>Qualified Operator</u>. Seller shall itself operate the Facility or cause the Facility to be operated by an entity that has at least two years of experience in operation of geothermal energy facilities of comparable size to the Facility and approved by PacifiCorp, which approval shall not be unreasonably withheld or delayed.

6.1.3 <u>Fines and Penalties</u>.

(a) Without limiting a Party's rights under Section 6.1.3(b), each Party shall pay all fines and penalties incurred by such Party on account of noncompliance by such Party with Requirements of Law in respect to this Agreement, except where such fines and penalties are being contested in good faith through appropriate proceedings.

(b) If fines, penalties, or legal costs are assessed against or incurred by either Party (the "Indemnified Party") on account of any action by any Governmental Authority due to noncompliance by the other Party (the "Indemnifying Party") with any Requirements of Law or the provisions hereof, or if the performance of the Indemnifying Party is delayed or stopped by order of any Governmental Authority due to the Indemnifying Party's noncompliance with any Requirements of Law, the Indemnifying Party shall indemnify and hold harmless the Indemnified Party against any and all losses, liabilities, damages, and claims suffered or incurred by the Indemnified Party as a result thereof. Without limiting the generality of the foregoing, the Indemnifying Party shall reimburse the Indemnified Party for all fees, damages, or penalties imposed on the Indemnified Party by any Governmental Authority, other person or to other utilities for violations to the extent caused by a default by the Indemnifying Party or a failure of performance by the Indemnifying Party hereunder.

6.2 <u>Interconnection</u>. Seller shall be responsible for the costs and expenses associated with interconnection of the Facility at its Nameplate Capacity Rating to and at the Point of Interconnection. Seller shall have no claims hereunder against PacifiCorp, acting in its merchant function capacity, with respect to any requirements imposed by or damages caused by (or allegedly caused by) acts or omissions of the Transmission Provider or Interconnection Provider, in connection with the Generation Interconnection Agreement or otherwise.

6.3 <u>Coordination with System</u>. Seller shall be responsible for the coordination and synchronization of the Facility and the Interconnection Facilities with the System. In the event there are unanticipated changes in FERC or Electric System Authority rules sufficiently significant to change the benefits, risks and burdens held by the Parties, the Parties shall meet in good faith to adjust the terms of this Agreement to provide for the Parties the originally intended allocation of benefits, risks and burdens.

6.4 <u>Outages</u>.

6.4.1 <u>Planned Outages</u>. Without the advance written approval of PacifiCorp, Seller shall not schedule a Planned Outage during any portion of any of the months of May, June, July, August, or September, except to the extent a Planned Outage during any such month is reasonably required to satisfy a manufacturer's warranty requirement or to comply with Prudent Electrical Practices. At least seven months prior to the Delivery Start Date, and at least seven months prior to the beginning of each Contract Year thereafter, Seller shall provide PacifiCorp, in writing, Seller's proposed schedule of Planned Outages for each month of the 18month period beginning on such Delivery Start Date or beginning of such Contract Year. To the extent Seller believes it is required to change any Planned Outage schedule in order to comply with Prudent Electrical Practices, it shall notify PacifiCorp of the same as soon as practicable and seek consent to such change, which consent shall not be unreasonably withheld, delayed or conditioned.

6.4.2 Maintenance Outages. If Seller reasonably determines that it is necessary to schedule a Maintenance Outage, Seller shall notify PacifiCorp of the proposed Maintenance Outage as soon as practicable but in any event at least 30 days before the outage begins (or such shorter period to which PacifiCorp may reasonably consent in light of thenexisting conditions). Upon such notice, the Parties shall plan the Maintenance Outage to mutually accommodate the reasonable requirements of Seller and the service obligations of PacifiCorp; provided, however, that Seller shall take all reasonable measures consistent with Prudent Electrical Practices not to schedule any Maintenance Outage during the months of May, June, July, August, or September Notice of a proposed Maintenance Outage shall include the expected start date and time of the outage, the amount of generation capacity of the Facility that will not be available, and the expected completion date and time of the outage. Seller shall give PacifiCorp notice of the Maintenance Outage as soon as practicable after Seller determines that a Maintenance Outage is necessary. PacifiCorp shall promptly respond to such notice and may request reasonable modifications in the schedule for the outage. Seller shall use all reasonable efforts to comply with any request to modify the schedule for a Maintenance Outage. Seller shall notify PacifiCorp of any subsequent changes in generation capacity available to PacifiCorp as a result of such Maintenance Outage or any changes in the Maintenance Outage completion date and time. As soon as practicable, any notifications given orally shall be confirmed in writing. Seller shall take all reasonable measures consistent with Prudent Electrical Practices to minimize the frequency and duration of Maintenance Outages.

6.4.3 <u>Forced Outages</u>. Seller shall promptly provide PacifiCorp an oral report, via telephone to a number specified by PacifiCorp, of any Forced Outage resulting in more than 10 percent of the Nameplate Capacity Rating of the Facility being unavailable. This report shall include the amount of the generation capacity of the Facility that will not be available because of the Forced Outage and the expected return date of such generation capacity. Seller shall promptly update the report as necessary to advise PacifiCorp of changed circumstances. As soon as practicable, the oral report shall be confirmed in writing by notice to PacifiCorp. Seller shall take all reasonable measures consistent with Prudent Electrical Practices to avoid Forced Outages and to minimize their duration.

6.4.4 <u>Notice of Deratings and Outages</u>. Without limiting the foregoing, Seller will inform PacifiCorp, via telephone to a number specified by PacifiCorp, of any major limitations, restrictions, deratings or outages known to Seller affecting the Facility for the following day (except for Curtailments) and will promptly update Seller's notice to the extent of any material changes in this information, with "major" defined as affecting more than 5 percent of the Nameplate Capacity Rating of the Facility.

6.4.5 <u>Effect of Outages on Estimated Output</u>. Seller represents and warrants that the estimated monthly Net Output set forth on Exhibit A takes into account the Planned Outages, Maintenance Outages, and Forced Outages that Seller reasonably expects to encounter in an average year in the ordinary course of operating the Facility.

6.5 <u>Scheduling</u>.

<u>Cooperation and Standards</u>. With respect to any and all scheduling requirements hereunder, (a) Seller shall cooperate with PacifiCorp with respect to scheduling Net Output, and (b) each Party shall designate authorized representatives to communicate with regard to scheduling and related matters arising hereunder. Each Party shall comply with the applicable variable resource standards and criteria of any applicable Electric System Authority.

6.6 <u>Forecasting</u>.

6.6.1 <u>Long-Range Forecasts</u>. For planning purposes, Seller shall, by December 1 of each year during the Term (except for the last year of the Term), provide an annual update to the expected long-term monthly/diurnal mean net capacity factor estimates (12 X 24 profile) by updating Exhibit A.

6.6.2 <u>Day-Ahead Forecasts and Updates</u>. By such time as mutually agreed to by the Parties on the Business Day immediately preceding the day on which Net Output from the Facility is to be delivered, Seller shall provide PacifiCorp with an hourly forecast of deliveries for each hour of the next day; provided, however, that a forecast provided on a day before any non-Business Day shall include forecasts for each day to and including the next Business Day. The Parties shall cooperate to implement and use automatic forecast updates including more frequent updates when material conditions change, including weather. Seller shall communicate forecasts under this Section 6.6.2 in an efficient manner, including electronic mail or other such media as determined by PacifiCorp (which, at PacifiCorp's discretion, may be in lieu of or in addition to notice to PacifiCorp).

6.6.3 <u>Basis of Forecasts</u>. The forecasts called for by this Agreement shall be non-binding, good faith estimates only, and PacifiCorp expressly releases and holds harmless Seller from any liability for forecasting errors.

6.6.4 <u>Compliance</u>. With respect to any and all forecasting requirements hereunder, and including real-time forecasting requirements, each Party shall comply with all applicable Electric System Authority tariff procedures, protocol, rules and testing as necessary. The Parties shall agree upon reasonable changes to these requirements from time-to-time as necessary to comply with the Electric System Authority, including but not limited to, automated forecasts and outage submissions. 6.7 Increase in Nameplate Capacity Rating; New Project Expansion or Development. Without limiting any restrictions herein on Nameplate Capacity Rating, if Seller elects to increase, at its own expense, the ability of the Facility to deliver Net Output in quantities in excess of the Maximum Delivery Rate through any means, including replacement or modification of turbines or related infrastructure, PacifiCorp shall not be required to purchase any Net Output above the Maximum Delivery Rate. If Seller or any Affiliate elects to build an expansion or additional geothermal project in the geographic vicinity of the Facility, Seller shall have no rights pursuant hereto to require PacifiCorp to purchase (and PacifiCorp shall have no obligation to purchase pursuant hereto) the output of any such expansion or additional facility. Any such expansion or additional facility may not materially and adversely impact the ability of either Party to fulfill its obligations pursuant hereto.

6.8 <u>Electronic Communications</u>.

6.8.1 <u>Telemetering</u>. Seller shall during the Term provide telemetering equipment and facilities capable of transmitting the following information concerning the Facility to PacifiCorp on a real-time basis, and operate such equipment when requested by PacifiCorp to indicate:

- (a) MW Output at the Point of Interconnection;
- (b) Net Output; and
- (c) the Facility's total instantaneous generation capacity.

Seller shall transmit to PacifiCorp any other data from the Facility that Seller receives on a real time basis. Seller shall provide such real time data to PacifiCorp on the same basis on which Seller receives the data (e.g., if Seller receives the data in four second intervals, PacifiCorp shall also receive the data in four second intervals).

6.8.2 <u>Transmission Provider Consent</u>. Seller shall execute a consent, in the form required by Transmission Provider, to provide that PacifiCorp can read the meter and receive any and all data from the Transmission Provider relating to transmission of Output or other matters relating to the Facility without the need for further consent from Seller.

6.8.3 <u>Dedicated Communication Circuit</u>. Seller shall install a dedicated direct communication circuit (which may be by common carrier telephone) between PacifiCorp and the control center in the Facility's control room or such other communication equipment as the Parties may agree.

6.9 <u>Reports and Records</u>.

6.9.1 <u>Monthly Reports</u>. Within 30 days after the end of each calendar month during the Term (each, a "Reporting Month"), Seller shall provide to PacifiCorp a report in electronic format, which report shall include (a) summaries of the Facility's Output data for the Reporting Month in intervals not to exceed one hour (or such shorter period as is reasonably possible with commercially available technology), including information from the Facility's computer monitoring system; (b) summaries of any other significant events related to the

operation of the Facility for the Reporting Month; and (c) any supporting information that PacifiCorp may from time to time reasonably request (including historical data for the Facility).

6.9.2 <u>Electronic Fault Log</u>. Seller shall maintain an electronic fault log of operations of the Facility during each hour of the Term. Seller shall provide PacifiCorp with a copy of the electronic fault log within 30 days after the end of the calendar month to which the fault log applies.

6.9.3 <u>Other Information to be Provided to PacifiCorp</u>. Seller shall provide to PacifiCorp the following information concerning the Facility:

(a) Upon the request of PacifiCorp, the manufacturers' guidelines and recommendations for maintenance of the Facility equipment;

(b) A report summarizing the results of maintenance performed during each Maintenance Outage, Planned Outage, and any Forced Outage, and upon request of PacifiCorp any of the technical data obtained in connection with such maintenance; and

(c) A monthly report detailing the Availability of the Facility.

6.9.4 <u>Information to Governmental Authorities</u>. Seller shall, promptly upon written request from PacifiCorp, provide PacifiCorp with all data collected by Seller related to the Facility reasonably required by PacifiCorp or an Affiliate thereof for reports to, and information requests from, any Governmental Authority or Electric System Authority. Along with this information, Seller shall provide to PacifiCorp copies of all submittals to Governmental Authorities or Electric System Authorities directed by PacifiCorp and related to the operation of the Facility with a certificate that the contents of the submittals are true and accurate to the best of Seller's knowledge. Seller shall use best efforts to provide this information to PacifiCorp with sufficient advance notice to enable PacifiCorp to review such information and meet any submission deadlines imposed by the requesting organization or entity. PacifiCorp shall reimburse Seller for all of Seller's reasonable costs and expenses in excess of \$10,000, if any, incurred in connection with PacifiCorp's requests for information under this Section 6.9.4.

6.9.5 <u>Data Request</u>. Seller shall, promptly upon written request from PacifiCorp, provide PacifiCorp with data reasonably required for information requests from any Governmental Authorities, state or federal agency intervener or any other party achieving intervenor status in any PacifiCorp rate proceeding or other proceeding before any Governmental Authority. Seller shall use best efforts to provide this information to PacifiCorp sufficiently in advance to enable PacifiCorp to review it and meet any submission deadlines. PacifiCorp shall reimburse Seller for all of Seller's reasonable costs and expenses in excess of \$10,000, if any, incurred in connection with PacifiCorp's requests for information under this Section 6.9.5.

6.9.6 <u>Documents to Governmental Authorities</u>. After sending or filing any statement, application, and report or any document with any Governmental Authority or Electric System Authority relating to operation and maintenance of the Facility, Seller shall promptly provide to PacifiCorp a copy of the same.

6.9.7 Environmental Information. Seller shall, promptly upon written request from PacifiCorp, provide PacifiCorp with all data reasonably requested by PacifiCorp relating to environmental information under the Required Facility Documents. Seller shall further provide PacifiCorp with information relating to environmental impact mitigation measures it is taking in connection with the Facility's construction or operation that are required by any Governmental Authority. PacifiCorp shall reimburse Seller for all of Seller's reasonable costs and expenses in excess of \$10,000, if any, incurred in connection with PacifiCorp's requests for the foregoing information under this Section 6.9.7. As soon as it is known to Seller, Seller shall disclose to PacifiCorp the extent of any material violation of any environmental laws or regulations arising out of the construction or operation of the Facility, or the presence of Environmental Contamination at the Facility or on the Premises, alleged to exist by any Governmental Authority having jurisdiction over the Premises, or the present existence of, or the occurrence during Seller's occupancy of the Premises of, any enforcement, legal, or regulatory action or proceeding relating to such alleged violation or alleged presence of Environmental Contamination presently occurring or having occurred during the period of time that Seller has occupied the Premises.

6.9.8 <u>Operational Reports</u>. Seller shall provide PacifiCorp monthly operational reports in a form and substance reasonably acceptable to PacifiCorp, and Seller shall, promptly upon written request from PacifiCorp, provide PacifiCorp with all operational data requested by PacifiCorp with respect to the performance of the Facility and delivery of Net Output, Capacity Rights therefrom.

6.9.9 <u>Notice of Material Adverse Events</u>. Seller shall promptly notify PacifiCorp of receipt of written notice or actual knowledge by Seller or its Affiliates of the occurrence of any event of default under any material agreement to which Seller is a party and of any other development, financial or otherwise, which would have a material adverse effect on Seller, the Facility or Seller's ability to operate, maintain or own the Facility as provided herein.

6.9.10 <u>Notice of Litigation</u>. Following its receipt of written notice or actual knowledge of the commencement of any action, suit, or proceeding before any court or Governmental Authority against Seller or its members with respect to this Agreement or the transactions contemplated hereunder, Seller shall promptly give notice to PacifiCorp of the same. Following its receipt of written notice or actual knowledge of the commencement of any action, suit or proceeding before any court or Governmental Authority against Seller, its members or any Affiliate, the effect of which would materially and adversely affect Seller's performance of its obligations hereunder, Seller shall promptly give notice to PacifiCorp of the same.

6.9.11 <u>Additional Information</u>. Seller shall provide to PacifiCorp such other information respecting the condition or operations of Seller, as such pertains to Seller's performance of its obligations hereunder, or the Facility as PacifiCorp may, from time to time, reasonably request.

6.9.12 <u>Confidential Treatment</u>. The monthly reports and other information provided to PacifiCorp under this Section 6.9 shall be treated as Confidential Business Information, subject to PacifiCorp's rights to disclose such information pursuant to Sections 6.9.4, 6.9.5, 9.5, and 24.3, and pursuant to any applicable Requirements of Law. Seller shall have the right to seek confidential treatment of any such information from the Governmental Authority entitled to receive such information.

6.10 <u>Financial and Accounting Information</u>. If PacifiCorp or one of its Affiliates determines that, under the Financial Accounting Standards Board's revised Interpretation No. 46, Consolidation of Variable Interest Entities ("FIN 46"), it may hold a variable interest in Seller, but it lacks the information necessary to make a definitive conclusion, Seller hereby agrees to provide, upon PacifiCorp's written request, sufficient financial and ownership information so that PacifiCorp or its Affiliate may confirm whether a variable interest does exist under FIN 46. If PacifiCorp or its Affiliate determines that, under FIN 46, it holds a variable interest in Seller, Seller hereby agrees to provide, upon PacifiCorp's written request, sufficient financial and other information to PacifiCorp or its Affiliate so that PacifiCorp may properly consolidate the entity in which it holds the variable interest or present the disclosures required by FIN 46. PacifiCorp shall reimburse Seller for Seller's reasonable costs and expenses, if any, incurred in connection with PacifiCorp's requests for information under this Section 6.10.

6.11 <u>Output Guarantee</u>.

6.11.1 Guaranteed Energy. Seller acknowledges that University is relying upon the full quantity of Expected Energy each Contract Year. Seller promises and guarantees that it will deliver from the Facility to PacifiCorp for redelivery to the University hereunder at least of the Expected Energy as specified in Exhibit A each Contract Year, excluding Energy that is not delivered during a Seller Excused Delivery Hour ("Guaranteed Energy"). The calculation of the number of MWhs related to each day of a Planned Outage, an unplanned outage, or the seven other days related to a Seller Excused Delivery Hour will be based on the average daily MWh of the month in which the outage occurred as set forth in Schedule A. Seller will be liable for agreed-upon liquidated damages ("Liquidated Damages") for failure of any Guaranteed Energy to be delivered to PacifiCorp for redelivery to University in any Contract Year, in the below the Guaranteed Energy actually delivered from the Facility amount of to PacifiCorp for redelivery to University in a Contract Year. Each Party agrees and acknowledges that the damages that would be incurred due to Seller's failure to deliver the Guaranteed Energy from the Facility each Contract Year would be difficult or impossible to predict with certainty and that the Liquidated Damages contemplated by this provision represent a fair and reasonable calculation of such damages.

6.11.2 <u>Guaranteed Power</u>. Seller acknowledges that University is relying upon the delivery of Power from the Facility during each On-Peak Hour during each Contract Year. Seller guarantees that it will deliver Power from, and commensurate with Net Output of, the Facility in each such On-Peak Hour, except during a Seller Excused Delivery Hour ("Guaranteed Power"). If Guaranteed Power is not delivered during any On-Peak Hour in any day, Seller will be liable for Liquidated Damages for each Disruption Day in the amount of **Deliver**. Each Party agrees and acknowledges that the damages that would be incurred due to Seller's failure to deliver Guaranteed Power from the Facility each On-Peak Hour would be difficult or impossible to predict with certainty and that the Liquidated Damages contemplated by this provision represent a fair and reasonable calculation of such damages. 6.11.3 Damages Invoicing. No later than 30 days after the end of each contract month or year, as applicable, Seller shall provide to PacifiCorp and University its calculation of and records supporting Liquidated Damages under this Section 6.11. No later than 60 days following the end of each month, PacifiCorp shall deliver to Seller an invoice for any Liquidated Damages. In preparing such calculations, Parties shall utilize meter data for the month in question, but may rely on historical averages and such other information as may be available at the time if meter data is incomplete or not available. To the extent required, any such invoice shall be trued up as promptly as practicable following receipt of actual results. Seller shall pay to PacifiCorp, by wire transfer of immediately available funds to an account specified in writing by PacifiCorp or by any other means agreed to by the Parties in writing from time to time, the amount set forth as due in such invoice, and shall within 30 days after receiving the invoice raise any objections regarding any disputed portion of the invoice. All disputes regarding such invoices shall be subject to Section 10.4. Objections not made by Seller within the 30-day period shall be deemed waived.

6.12 <u>Access Rights</u>. Upon reasonable prior notice and subject to the prudent safety requirements of Seller, and Requirements of Law relating to workplace health and safety, Seller shall provide PacifiCorp and its authorized agents, employees and inspectors ("PacifiCorp Representatives") with reasonable access to the Facility: (a) for the purpose of reading or testing metering equipment, (b) as necessary to witness any acceptance tests, (c) to provide tours of the Facility to customers and other guests of PacifiCorp (not more than 12 times per year), (d) for purposes of implementing Sections 2.3 or 10.5, and (e) for other reasonable purposes at the reasonable request of PacifiCorp. PacifiCorp shall release Seller against and from any and all Liabilities resulting from actions or omissions by any of the PacifiCorp Representatives in connection with their access to the Facility, except to the extent that such damages are caused or by the intentional or grossly negligent act or omission of Seller.

6.13 <u>Facility Images</u>. PacifiCorp shall be free to use any and all images from or of the Facility for promotional purposes, subject to Seller's consent (not to be unreasonably withheld or delayed, and which consent may consider Requirements of Law relating to Premises security, obligations to outside vendors (including any confidentiality obligations), and the corporate policies of Seller's Affiliates).

SECTION 7 FACILITY STATUS

Seller shall provide PacifiCorp with copies of Seller's FERC Exempt Wholesale Generator status and for authority to sell energy. During the Term, Seller shall maintain its status (to the extent it is required by law to do so) and its authority to sell power hereunder.

SECTION 8 SECURITY

8.1 <u>Project Development Security</u>. Project Development Security requirements related to this Facility are set forth in the Renewable Energy Supply Agreement. PacifiCorp shall have no obligation or right to require, enforce or draw upon any Project Development Security associated with the Facility.

8.2 <u>Default Security</u>. On or before the Delivery Start Date, Seller shall post and maintain, or cause to be posted and maintained, with and for the benefit of PacifiCorp and University, security in the form of cash or a Letter of Credit in the amount of secure performance of all of Seller's obligations under this Agreement. If Seller fails to timely discharge any of its obligations under this Agreement, PacifiCorp may draw upon the Default Security on account of any resulting damages or University Damages. PacifiCorp shall also be entitled to draw upon the Default Security for damages if this Agreement is terminated because of an Event of Default by Seller. In the event University cures any default by Seller or enforces its third-party beneficiary rights under this Agreement, University may thereafter draw upon any Default Security and apply the proceeds against any damages.

8.3 <u>No Limitation</u>. The security required hereunder (a) constitutes security for, but not a limitation on, Seller's obligations under this Agreement and (b) shall not be an exclusive remedy for Seller's failure to perform in accordance with this Agreement. To the extent any Default Security is drawn upon, Seller shall, on or before the first day of the month following such draw, replenish or reinstate the security to the full amount required under this Agreement

SECTION 9 METERING

9.1 <u>Installation of Metering Equipment</u>. Metering equipment shall be designed, furnished, installed, owned, inspected, tested, maintained and replaced as provided in the Generation Interconnection Agreement at the Point of Interconnection and as required by Transmission Provider; provided, however, that PacifiCorp acting in its merchant function capacity shall be under no obligation, pursuant hereto, to bear any expense relating to any such metering equipment. If PacifiCorp desires metering above and beyond that as specified above, PacifiCorp shall be responsible for all associated costs.

9.2 <u>Metering</u>. Metering shall be performed at the location and in the manner specified in Exhibit I, the Generation Interconnection Agreement and as necessary to perform and Seller's other obligations hereunder. All quantities of Net Output purchased hereunder shall reflect the net amount of energy flowing into the System at the Point of Delivery.

9.3 Inspection, Testing, Repair and Replacement of Meters. PacifiCorp shall have the right to periodically inspect and request the interconnecting utility, test, repair and replace the metering equipment at the Point of Interconnection or the Point of Delivery, without PacifiCorp assuming any obligations thereunder. If any of the inspections or tests disclose an error exceeding 0.5 percent, either fast or slow, proper correction, based upon the inaccuracy found, shall be made of previous readings for the actual period during which the metering equipment rendered inaccurate measurements if that period can be ascertained. If the actual period cannot be ascertained, the proper correction shall be made to the measurements taken during the time the metering equipment was in service since last tested, but not exceeding three months, in the amount the metering equipment shall have been shown to be in error by such test. Any correction in billings or payments resulting from a correction, when made, shall constitute full adjustment of any claim between Seller and PacifiCorp arising out of such inaccuracy of metering equipment. Nothing in this Agreement shall give rise to PacifiCorp, acting in its

merchant function capacity hereunder, having any obligations to Seller, or any other person or entity, pursuant to or under the Generation Interconnection Agreement.

9.4 <u>Metering Costs</u>. To the extent not otherwise provided in the Generation Interconnection Agreement, Seller shall bear all costs relating to all metering equipment installed to accommodate Seller's Facility.

9.5 <u>Meter Data</u>. Within 10 days of the Effective Date, Seller shall request the Interconnection Provider or Transmission Provider in writing to provide any and all meter or other data associated with the Facility or Net Output directly to PacifiCorp. Notwithstanding any other provision hereof, PacifiCorp shall have the right to provide such data to any Electric System Authority.

9.6 <u>WREGIS Metering</u>. Seller shall cause the Facility to implement all necessary generation information communications in WREGIS, and report generation information to WREGIS pursuant to a WREGIS-approved meter that is dedicated to the Facility and only the Facility. Seller shall at its sole cost and expense indemnify and hold PacifiCorp harmless from all costs and expenses incurred by PacifiCorp in connection with any dispute between Seller and WREGIS or WECC in connection with WREGIS, including responding to data requests or subpoenas, and including taking such steps as are necessary to cause the facility to generate WREGIS Certificates in WREGIS should WREGIS refuse to do business with Seller.

SECTION 10 BILLINGS, COMPUTATIONS AND PAYMENTS

10.1 <u>Monthly Invoices</u>. On or before the 10th day following the end of each calendar month, Seller shall deliver to PacifiCorp a proper invoice showing Seller's computation of Net Output delivered to the Point of Delivery during such month. If such invoice is delivered by Seller to PacifiCorp, then PacifiCorp shall send to Seller, on or before the later of the 20th day following receipt of such invoice or the 30th day following the end of each month, payment for Seller's deliveries of Net Output to PacifiCorp.

10.2 <u>Offsets</u>. Either Party may offset any payment due hereunder against amounts owed by the other Party pursuant hereto. Either Party's exercise of recoupment and set off rights shall not limit the other remedies available to such Party hereunder.

10.3 <u>Interest on Late Payments</u>. Any amounts that are not paid when due hereunder shall bear interest at the Contract Interest Rate from the date due until paid.

10.4 <u>Disputed Amounts</u>. If either Party, in good faith, disputes any amount due pursuant to an invoice rendered hereunder, such Party shall notify the other Party of the specific basis for the dispute and, if the invoice shows an amount due, shall pay that portion of the statement that is undisputed, on or before the due date. Any such notice shall be provided within two years of the date of the invoice in which the error first occurred. If any amount disputed by such Party is determined to be due the other Party, or if the Parties resolve the payment dispute, the amount due shall be paid within five (5) days after such determination or resolution, along with interest at the Contract Interest Rate from the date due until the date paid.

10.5 <u>Audit Rights</u>. Each Party, through its authorized representatives, shall have the right, at its sole expense upon reasonable notice and during normal business hours, to examine and copy the records of the other Party to the extent reasonably necessary to verify the accuracy of any statement, charge or computation made hereunder or to verify the other Party's performance of its obligations hereunder. Upon request, each Party shall provide to the other Party statements evidencing the quantities of Net Output delivered at the Point of Delivery. If any statement is found to be inaccurate, a corrected statement shall be issued and any amount due thereunder will be promptly paid and shall bear interest at the Contract Interest Rate from the date of the overpayment or underpayment to the date of receipt of the reconciling payment. Notwithstanding the foregoing, no adjustment shall be made with respect to any statement or payment hereunder unless a Party questions the accuracy of such payment or statement within two years after the date of such statement or payment.

SECTION 11 DEFAULTS AND REMEDIES

11.1 <u>Defaults</u>. The following events are defaults (each a "default" before the passing of applicable notice and cure periods, and an "Event of Default" thereafter) hereunder:

11.1.1 Defaults by Either Party.

(a) A Party fails to make a payment when due hereunder if the failure is not cured within 15 Business Days after the non-defaulting Party gives the defaulting Party a notice of the default.

(b) A Party (i) makes an assignment for the benefit of its creditors; (ii) files a petition or otherwise commences, authorizes or acquiesces in the commencement of a proceeding or cause of action under any bankruptcy or similar law for the protection of creditors, or has such a petition filed against it and such petition is not withdrawn or dismissed within 60 days after such filing; (iii) becomes insolvent; or (iv) is unable to pay its debts when due.

(c) A Party breaches a representation or warranty made by it herein if the breach is not cured within 30 days after the non-defaulting Party gives the defaulting Party a notice of the default; provided that if such default is not reasonably capable of being cured within the thirty (30) day cure period but is reasonably capable of being cured within an additional one hundred and eighty (180) day cure period, the defaulting Party will have such additional time (not exceeding an additional 180 days) as is reasonably necessary to cure, if, prior to the end of the 30 day cure period the defaulting Party provides the non-defaulting Party a remediation plan, the non-defaulting party approves such remediation plan, which approval shall not be unreasonably withheld or delayed, and the defaulting Party promptly commences and diligently pursues the remediation plan. 11.1.2 A Party otherwise fails to perform any material obligation hereunder for which an exclusive remedy is not provided hereunder and which is not addressed in any other Event of Default described in Section 11.1, if the failure is not cured within 30 days after the non-defaulting Party gives the defaulting Party notice of the default; provided that if such default is not reasonably capable of being cured within the thirty (30) day cure period but is reasonably capable of being cured within an additional one hundred and eighty (180) day cure period, the defaulting Party will have such additional time (not exceeding an additional 180 days) as is reasonably necessary to cure, if, prior to the end of the 30 day cure period the defaulting Party provides the non-defaulting Party a remediation plan, the non-defaulting party approves such remediation plan, which approval shall not be unreasonably withheld or delayed, and the defaulting Party promptly commences and diligently pursues the remediation plan.

11.1.3 Defaults by Seller.

(a) Seller fails to post, increase, or maintain the Default Security as required hereunder by the applicable dates set forth therein.

(b) Seller sells Net Output, Capacity Rights or RECs from the Facility to a party other than PacifiCorp or University, in breach hereof.

(c) PacifiCorp receives notice of foreclosure of the Facility or any part thereof by a Lender to Seller, mechanic or materialman, or any other holder, of an unpaid lien or other charge or encumbrance, if the same has not been stayed, paid, or bonded around within 30 days of the date of the notice received by PacifiCorp, and in all cases prior to the expiration of any applicable cure period associated with such foreclosure.

(d) Seller fails to maintain any Required Facility Documents or Permits necessary to own or operate the Facility, after the expiration of applicable notice, cure and waiver periods.

(e) Seller's Abandonment of construction or operation of the Facility except to the extent caused by an event of Force Majeure or default by PacifiCorp.

(f) Seller fails to maintain insurance as required by the Agreement.

11.2 <u>Remedies for Failure to Deliver/Receive</u>.

11.2.1 <u>Remedy for Seller's Failure to Deliver</u>. Upon the occurrence and during the continuation of a default of Seller under Section 11.1.3(b), Seller shall pay PacifiCorp within five (5) Business Days after invoice receipt, an amount equal to (a) PacifiCorp's Cost to Cover multiplied by the Net Output delivered to a party other than PacifiCorp, (b) additional costs and transmission charges, if any, reasonably incurred by PacifiCorp in moving replacement energy to the Point of Delivery or if not there, to such points in PacifiCorp's control area as are determined by PacifiCorp, (c) any additional cost or expense incurred as a result of Seller's default under Section 11.1.3(b), as determined by PacifiCorp in a commercially reasonable manner (but not including any penalties, ratcheted demand or similar charges), and (d) University Damages. The invoice for such amount shall include a written statement explaining in reasonable detail the calculation of such amount.

11.2.2 <u>Remedy for PacifiCorp's Failure to Purchase</u>. If PacifiCorp fails to receive or purchase all or part of the Net Output required to be purchased pursuant hereto and such failure is not excused under the terms hereof or by Seller's failure to perform, then Seller shall perform under Section 11.7 and PacifiCorp shall pay Seller, on the earlier of the date payment would otherwise be due in respect of the month in which the failure occurred or within five (5) Business Days after invoice receipt, an amount equal to Seller's Cost to Cover multiplied by the amount of Net Output so not received or purchased, less amounts received by Seller pursuant to Section 11.7, plus all costs and expenses incurred by Seller in arranging for sales to third-parties, including the costs and expenses related to entering into an alternative transaction and any costs in delivering Net Output. The invoice for such amount shall include a written statement explaining in reasonable detail the calculation thereof.

11.2.3 <u>Remedy for Seller's Failure to Sell/Deliver Capacity Rights</u>. Seller shall be liable for PacifiCorp's actual damages in the event Seller fails to sell or deliver all or any portion of the Capacity Rights to PacifiCorp.

11.3 <u>Remedies for University Default</u>. If University defaults in its obligation to make required payments to PacifiCorp under the University Contract and the University fails to cure such payment default within the applicable cure period specified therein, PacifiCorp's obligation to purchase further Net Output under this Agreement may be suspended by PacifiCorp until University remedies such default, effective upon delivery of written notice of such suspension by PacifiCorp to Seller. If University fails to cure such default within the applicable cure period, PacifiCorp's obligation to acquire or purchase Net Output under this Agreement shall cease and, subject to notice and termination provisions herein, PacifiCorp may elect to terminate this Agreement.

Termination and Remedies. Promptly after a non-defaulting Party has knowledge 11.4 of any default, it shall provide notice to the defaulting Party, Lender and University, specifying in reasonable detail the default and the cure period, if any, applicable thereto. From and during the continuance of an Event of Default, the non-defaulting Party shall be entitled to all remedies available at law or in equity, and may terminate this Agreement by notice to the other Party designating the date of termination and delivered to the defaulting Party no less than 60 days before such termination date; provided, however, that as a precondition to Seller's exercise of this termination right, Seller must also provide copies of such notice to the notice addresses of then-current President and General Counsel of PacifiCorp set forth in Section 23. Also, as a precondition to PacifiCorp's exercise of this termination right, PacifiCorp must also provide copies of such notice to the notice addresses of Seller, Lender and University set forth in Section 23. Such copies shall be sent by registered overnight delivery service or by certified or registered mail, return receipt requested. In addition, a Party's termination notice shall state prominently therein in type font no smaller than 14-point all-capital letters that "THIS IS A TERMINATION NOTICE UNDER A RENEWABLE PPA. YOU MUST CURE A DEFAULT, OR THE PPA MAY BE TERMINATED," and shall state therein any amount purported to be owed and wiring instructions. A Party will not have any right to terminate this Agreement if the default that gave rise to the termination right is cured within 60 days of receipt of such notice or is subject to the notice and cure rights set forth in this Section. Further, from and after the date upon which Seller fails to remedy a default within the time periods provided in Section 11.1, and until PacifiCorp has recovered all damages incurred on account of such default by Seller, without exercising its termination right, PacifiCorp may offset its damages against any payment due Seller. Except in circumstances in which a remedy provided for in this Agreement is described as a Party's sole or exclusive remedy, upon termination, the non-defaulting Party may pursue any and all legal or equitable remedies provided by law, equity or this Agreement. The rights contemplated by this Section 11 are cumulative such that the exercise of one or more rights shall not constitute a waiver of any other rights. In the event of a termination hereof:

(a) Each Party shall pay to the other all amounts due the other hereunder for all periods prior to termination, subject to offset by the non-defaulting Party against damages incurred by such Party.

(b) Any amounts due hereunder shall be calculated and paid within 30 days after the billing date for such charges and shall bear interest thereon at the Contract Interest Rate from the date of default until the date paid. The foregoing does not extend the due date of, or provide an interest holiday for any payments otherwise due hereunder.

(c) Before and after the effective date of termination, the nondefaulting Party may pursue, to the extent permitted by this Agreement, any and all legal or equitable remedies provided by law, equity or this Agreement.

(d) Without limiting the generality of the foregoing, the provisions of Sections 4.5, 5.4, 6.9.4, 6.9.5, 8, 9.5, 9.6, 10.3, 10.4, 10.5, 11.4, 11.5, 11.6, 11.10, and Section 13 and Section 24 shall survive the termination hereof.

11.5 <u>Termination of Duty to Buy</u>. If this Agreement is terminated because of a default by Seller, neither Seller nor any Affiliate of Seller, nor any successor to Seller with respect to the ownership of the Facility or Premises, on whose behalf Seller acts herein as agent, may thereafter require or seek to require PacifiCorp to make any purchases from the Facility or any electric generation facility constructed on the Premises under PURPA, or any other Requirements of Law, for any periods that would have been within the Term had this Agreement remained in effect. Seller, on behalf of itself and on behalf of any other entity on whose behalf it may act, hereby waives its rights to require PacifiCorp to do so. This section shall not apply to University, to the extent it cures Seller's default, succeeds to Seller's interest in the Facility, or otherwise obtains the right to continue purchasing renewable energy from the Facility.

11.6 <u>Termination Damages</u>. If this Agreement is terminated as a result of an Event of Default by one of the Parties, termination damages shall be determined. The amount of termination damages shall be calculated by the non-defaulting Party within a reasonable period after termination of the Agreement. Amounts owed pursuant to this section shall be due within five (5) Business Days after the non-defaulting Party gives the defaulting Party notice of the amount due. The non-defaulting Party shall under no circumstances be required to account for or otherwise credit or pay the defaulting Party for economic benefits accruing to the non-defaulting Party as a result of the defaulting Party's default.

11.7 <u>Duty/Right to Mitigate</u>. Each Party agrees that it has a duty to mitigate damages and covenants that it will use commercially reasonable efforts to minimize any damages it may incur as a result of the other Party's performance or non-performance hereof. "Commercially

reasonable efforts" (a) by Seller shall include requiring Seller to use commercially reasonable efforts to maximize the price for Net Output and RECs received by Seller from third parties, including entering into an enabling agreement with, or being affiliated with, one or more power marketers of nationally recognized standing to market such Net Output or RECs not purchased or accepted by PacifiCorp (only during a period PacifiCorp is in default), in each case only to the extent any of the foregoing actions are permitted under Requirements of Law and the Interconnection Agreement; and (b) by PacifiCorp shall include requiring PacifiCorp to use commercially reasonable efforts to minimize the price paid to third parties for energy purchased to replace Net Output and RECs not delivered by Seller as required hereunder.

11.8 <u>Security</u>. If this Agreement is terminated because of Seller's default, PacifiCorp may, in addition to pursuing any and all other remedies available at law or in equity, proceed against any security held by PacifiCorp or University in whatever form to reduce any amounts that Seller owes PacifiCorp arising from such default.

11.9 <u>Cure by Lender or University</u>.

(a) Seller shall notify PacifiCorp and University of the names and addresses of any Lender.

(b) PacifiCorp upon serving Seller any notice of default pursuant to any provisions of this Agreement, shall also serve a copy of such notice upon any Lender and University at the addresses provided for in Section 23. No notice of default by PacifiCorp hereunder shall be deemed to have been duly given unless and until a copy thereof shall have been so served.

(c) From and after the date that such notice has been given to a Lender and University, said Lender or University shall, each in its sole discretion, have the same period for remedying any alleged default or causing the same to be remedied, as is given to Seller pursuant to this Agreement. Lender or University may cure any default by either Party under this Agreement and recover from the defaulting Party any damages, costs or expenses suffered or incurred by Lender or University in enforcing this Agreement or curing a default, including reasonable attorneys' fees. The Parties shall accept such performance by or on behalf of such Lender or University as if the same had been done by Seller. If Seller is not proceeding with due diligence to remedy any such default, neither PacifiCorp nor Seller shall interpose any objection if any Lender or University takes such action for such purpose.

11.10 <u>Cumulative Remedies</u>. Except in circumstances in which a remedy provided for in this Agreement is described as a sole or exclusive remedy, the rights and remedies provided to PacifiCorp or Seller hereunder are cumulative and not exclusive of any rights or remedies of PacifiCorp or Seller.

11.11 <u>Recoupment of Damages</u>.

(a) <u>Default Security Available</u>. If Seller has posted Default Security, PacifiCorp may draw upon that security to satisfy any damages.

(b) <u>Recovery of Damages</u>. PacifiCorp may, with or without drawing upon any available Default Security, collect any amount owing by Seller hereunder by partially withholding future payments to Seller over a reasonable period of time, which period shall not be less than the period over which the default occurred. PacifiCorp and Seller shall work together in good faith to establish the period, and monthly amounts, of such withholding so as to avoid Seller's default on its commercial or financing agreements necessary for its continued operation of the Facility.

SECTION 12 EARLY TERMINATION

This Agreement and the obligations of the Parties hereunder may be terminated early by written notice from PacifiCorp to Seller, with no liability to either Party, upon receipt by PacifiCorp of written notice from University under the University Contract directing PacifiCorp to terminate this Agreement early as of a date specified by University ("Early Termination"), as set forth in Article IX of the Renewable Energy Supply Agreement between Seller and University.

SECTION 13 INDEMNIFICATION AND LIABILITY

13.1 Indemnities.

13.1.1 Indemnity by Seller. To the extent permitted by Requirements of Law and subject to Section 13.1.5, Seller shall release, indemnify and hold harmless PacifiCorp, its Affiliates, and each of its and their respective directors, officers, employees, agents, and representatives (collectively, the "PacifiCorp Indemnitees") against and from any and all losses, fines, penalties, claims, demands, damages, liabilities, actions or suits of any nature whatsoever (including legal costs and attorneys' fees, both at trial and on appeal, whether or not suit is brought) (collectively, "Liabilities") actually or allegedly resulting from, or arising out of, or in any way connected with, the performance by Seller of its obligations hereunder, for or on account of: (a) injury, bodily or otherwise, to, or death of, or (b) damage to, or destruction of property of, any person or entity, excepting only to the extent such Liabilities as may be caused by the gross negligence or willful misconduct of any person or entity within the PacifiCorp Indemnitees. Seller shall be solely responsible for (and shall defend and hold PacifiCorp harmless against) any damage that may occur as a direct result of Seller's breach of the Generation Interconnection Agreement.

13.1.2 <u>Indemnity by PacifiCorp</u>. To the extent permitted by Requirements of Law and subject to Section 13.1.5, PacifiCorp shall release, indemnify and hold harmless Seller, its Affiliates, and each of its and their respective directors, officers, employees, agents, and representatives (collectively, the "Seller Indemnitees") against and from any and all Liabilities actually or allegedly resulting from, or arising out of, or in any way connected with, the performance by PacifiCorp of its obligations hereunder for or on account of (a) injury, bodily or otherwise, to, or death of, or (b) for damage to, or destruction of property of, any person or entity, excepting only to the extent such Liabilities as may be caused by the gross negligence or willful misconduct of any person or entity within the Seller Indemnitees.

13.1.3 <u>Additional Cross Indemnity</u>. Without limiting Sections 13.1.1 and 13.1.2, Seller shall release, indemnify and hold harmless the PacifiCorp Indemnitees from and against all Liabilities related to Net Output prior to its delivery by Seller at the Point of Delivery, and PacifiCorp shall release, indemnify and hold harmless the Seller Indemnitees from and against all Liabilities related to Net Output once delivered to PacifiCorp at the Point of Delivery as provided herein, except in each case to the extent such Liabilities are attributable to the gross negligence or willful misconduct or a breach of this Agreement by any member of the PacifiCorp Indemnitees or the Seller Indemnitees, respectively, seeking indemnification hereunder.

13.1.4 <u>No Dedication</u>. Nothing herein shall be construed to create any duty to, any standard of care with reference to, or any liability to any person not a Party. No undertaking by one Party to the other under any provision hereof shall constitute the dedication of PacifiCorp's facilities or any portion thereof to Seller or to the public, nor affect the status of PacifiCorp as an independent public utility corporation or Seller as an independent individual or entity.

13.1.5 <u>Consequential Damages</u>. NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR SPECIAL, PUNITIVE, INDIRECT, EXEMPLARY OR CONSEQUENTIAL DAMAGES, WHETHER SUCH DAMAGES ARE ALLOWED OR PROVIDED BY CONTRACT, TORT (INCLUDING NEGLIGENCE), STRICT LIABILITY, STATUTE OR OTHERWISE. THE PARTIES AGREE THAT ANY LIQUIDATED DAMAGES, DELAY DAMAGES, PACIFICORP AND SELLER COST TO COVER OR UNIVERSITY DAMAGES, OR OTHER SPECIFIED MEASURE OF DAMAGES EXPRESSLY PROVIDED FOR HEREIN, ARE NOT INTENDED BY THEM TO REPRESENT SPECIAL, PUNITIVE, INDIRECT, EXEMPLARY OR CONSEQUENTIAL DAMAGES.

SECTION 14 INSURANCE

14.1 <u>Required Policies and Coverages</u>. Without limiting any liabilities or any other obligations of Seller hereunder, Seller shall secure and continuously carry with an insurance company or companies rated not lower than "A-VII" by the A.M. Best Company the insurance coverage specified on Exhibit J during the periods specified on Exhibit J.

14.2 <u>Certificates of Insurance</u>. Seller shall provide PacifiCorp with certificates of insurance within 10 days after the date by which such policies are required to be obtained (as set forth in Exhibit J). Seller shall provide a certificate of insurance (in ACORD or similar industry form) to PacifiCorp within 10 days of the effective date of any insurance policy required under this Agreement. Seller shall provide PacifiCorp with copies of any notices of cancellation that it may receive within five (5) Business Days of receipt from its insurer, and shall provide proof of replacement coverage prior to the effective date of cancellation. If any coverage is written on a "claims-made" basis, the certification accompanying the policy shall conspicuously state that the policy is "claims made."

SECTION 15 FORCE MAJEURE

15.1 <u>Definition of Force Majeure</u>. "Force Majeure" or "an event of Force Majeure" means an event that (a) is not reasonably anticipated as of the date hereof, (b) is not within the reasonable control of the Party affected by the event, (c) is not the result of such Party's negligence or failure to act, and (d) could not be overcome by the affected Party's use of due diligence in the circumstances.

15.2 <u>Examples of Force Majeure</u>. Events of Force Majeure include, but are not restricted to, events of the following types (but only to the extent that such an event, in consideration of the circumstances, satisfies the tests set forth in Section 15.1): (a) acts of God or nature such as fire, chemical or radioactive contamination or ionizing radiation, earthquakes, lightning, cyclones, hurricanes or floods; (b) civil disturbance; (c) sabotage; (d) strikes; lock-outs; work stoppages; and action or restraint by court order or public or government authority (as long as the affected Party has not applied for or assisted in the application for, and has opposed to the extent reasonable, such court or government action); (e) unanticipated and unforeseeable geological or ground conditions, and (f) specific incidents of exceptionally adverse temperature and wind speed conditions in excess of those necessary for the Facility to operate as designed and which are at least 25% worse than those encountered at the Facility at the relevant time of year during any of the 20 years prior to the Effective Date.

Exclusions from Force Majeure. None of the following constitute Force Majeure: 15.3 (i) Seller's ability to sell, or University or PacifiCorp's ability to purchase capacity, energy or RECs at a more advantageous price than is provided hereunder; (ii) the cost or availability of fuel or motive force to operate the Facility; (iii) economic hardship, including lack of money; (iv) any breakdown or malfunction of the Facility equipment (including any serial equipment defect) that is not caused by an independent event of Force Majeure, (v) the imposition upon a Party of costs or taxes allocated to such Party under Section 5, (vi) delay or failure of Seller to obtain or perform any Required Facility Document unless due to a Force Majeure event, (vii) any delay, alleged breach of contract, or failure by the Transmission Provider, Network Service Provider or Interconnection Provider unless due to a Force Majeure event, (viii) maintenance upgrade or repair of any facilities or right of way corridors constituting part of or involving the Interconnection Facilities, whether performed by or for Seller, or other third parties (except for repairs made necessary as a result of an event of Force Majeure); (ix) Seller's failure to obtain, or perform under, the Generation Interconnection Agreement, or its other contracts and obligations to Transmission Provider, Network Service Provider, or Interconnection Provider, unless due to a Force Majeure event; or (xi) any event attributable to the use of Interconnection Facilities for deliveries of Net Output to any party other than PacifiCorp. Notwithstanding anything to the contrary herein, in no event will the increased cost of electricity, steel, labor, or transportation constitute an event of Force Majeure.

15.4 <u>Suspension of Performance</u>. Neither Party shall be liable for any delay or failure in its performance under this Agreement, nor shall any delay, failure, or other occurrence or event become an Event of Default, to the extent such delay, failure, occurrence or event is substantially caused by conditions or events of Force Majeure duration of the continuation of the event of Force Majeure, for the same number of days that the event of Force Majeure has prevailed, provided that:

(a) the Party affected by the Force Majeure, shall, as soon as practicable after the occurrence of the event of Force Majeure, give the other Party and University written notice describing the particulars of the event; and

(b) the suspension of performance shall be of no greater scope and of no longer duration than is required to remedy the effect of the Force Majeure; and

to perform.

(c) the affected Party shall use diligent efforts to remedy its inability

15.5 <u>Force Majeure Does Not Affect Other Obligations</u>. No obligations of either Party that arose before the Force Majeure causing the suspension of performance or that arise after the

cessation of the Force Majeure shall be excused by the Force Majeure.

15.6 <u>Strikes</u>. Notwithstanding any other provision hereof, neither Party shall be required to settle any strike, walkout, lockout or other labor dispute on terms which, in the sole judgment of the Party involved in the dispute, are contrary to the Party's best interests.

15.7 <u>Right to Terminate</u>. If a Force Majeure event prevents a Party from substantially performing its obligations hereunder for a period exceeding 180 consecutive days, then the Party not affected by the Force Majeure event, with respect to its obligations hereunder, may terminate this Agreement by giving 10 days' prior notice to the other Party; provided, however, if the affected Party takes all reasonable steps to remedy the effects of the Force Majeure event with all reasonable dispatch, and the Force Majeure event cannot be resolved within 180 days, then this Agreement may not be terminated for 360 days. Notwithstanding the foregoing, unless University has failed to timely cure any University default under the University Contract, PacifiCorp may not terminate this Agreement under this Section 15.7 without the advance written consent of University. Upon such termination, neither Party will have any liability to the other with respect to period following the effective date of such termination; provided, however, that this Agreement will remain in effect to the extent necessary to facilitate the settlement of all liabilities and obligations arising hereunder before the effective date of such termination.

SECTION 16 SEVERAL OBLIGATIONS

Nothing contained herein shall be construed to create an association, trust, partnership or joint venture or to impose a trust, partnership or fiduciary duty, obligation or liability on or between the Parties.

SECTION 17 CHOICE OF LAW

This Agreement shall be interpreted and enforced in accordance with the laws of the State of Utah, applying any choice of law rules that may direct the application of the laws of another jurisdiction.

SECTION 18 PARTIAL INVALIDITY

The Parties do not intend to violate any laws governing the subject matter hereof. If any of the terms hereof are finally held or determined to be invalid, illegal or void as being contrary to any applicable law or public policy, all other terms hereof shall remain in effect. The Parties shall use best efforts to amend this Agreement to reform or replace any terms determined to be invalid, illegal or void, such that the amended terms (a) comply with and are enforceable under applicable law, (b) give effect to the intent of the Parties in entering hereinto, and (c) preserve the balance of the economics and equities contemplated by this Agreement in all material respects.

SECTION 19 NON-WAIVER

No waiver of any provision hereof shall be effective unless the waiver is set forth in a writing that (a) expressly identifies the provision being waived, and (b) is executed by the Party waiving the provision. A Party's waiver of one or more failures by the other Party in the performance of any of the provisions hereof shall not be construed as a waiver of any other failure or failures, whether of a like kind or different nature.

SECTION 20 GOVERNMENTAL JURISDICTION AND AUTHORIZATIONS

This Agreement is subject to the jurisdiction of those Governmental Authorities having control over either Party or this Agreement. During the Term, Seller shall maintain all Permits required, as applicable, for the operation or ownership of the Facility.

SECTION 21 SUCCESSORS AND ASSIGNS

21.1 <u>Restriction on Assignments</u>. Except as expressly provided in this Section 21, neither Party may assign this Agreement or any of its rights or obligations hereunder without the prior written consent of the other Party.

21.2 <u>Permitted Assignments</u>. Notwithstanding Section 21.1, either Party may, without the need for consent from the other Party (but with notice to the other Party, including the names of the assignees), (a) transfer, sell, pledge, encumber or assign this Agreement or the accounts, revenues or proceeds therefrom in connection with any financing or other financial arrangements; (b) transfer or assign this Agreement to an Affiliate of such Party; or (c) transfer or assign this Agreement to any party succeeding to all or substantially all of the assets or generating assets of such Party (and, with respect to Seller, Seller shall be required to transfer or assign this Agreement to any party succeeding to all or substantially all of the assets of Seller); provided, however, that Seller shall not transfer, sell, encumber or assign this Agreement or any interest herein to any Affiliate of PacifiCorp without the prior written consent of PacifiCorp. Except with respect to collateral assignments for financing purposes and also except as otherwise provided above in the immediately preceding sentence, in every assignment hereof, the assignee

must (x) agree in writing to be bound by the terms and conditions hereof, (y) possess the same or similar experience, and possess the same or better creditworthiness, as the assignor, and (z) except as provided in the next sentence, obtain the consent of the other Party hereof for the assignment. PacifiCorp may assign this Agreement in whole or in part without the consent of Seller to any person or entity in the event that PacifiCorp ceases to be a load-serving entity if its assignee meets the requirements of clauses (x) and (y) in the immediately preceding sentence, in which event PacifiCorp shall be released from liability hereunder. The Party seeking to assign or transfer this Agreement shall be solely responsible for paying all costs of assignment.

SECTION 22 ENTIRE AGREEMENT

This Agreement supersedes all prior agreements, proposals, representations, negotiations, discussions or letters, whether oral or in writing, regarding the subject matter hereof. No modification hereof shall be effective unless it is in writing and executed by both Parties.

SECTION 23 NOTICES

Addresses and Delivery Methods. All notices, requests, statements or payments 23.1 shall be made to the addresses set out below. In addition, copies of a notice of termination of this Agreement under Section 11 shall contain the information required by Section 11 and shall be sent to the then-current President and General Counsel of PacifiCorp, Lender and University. Notices required to be in writing shall be delivered by letter, facsimile or other tangible documentary form. Notice by hand delivery shall be deemed to have been given when received or hand delivered or upon attempted delivery if delivery is refused or is not reasonably possible because of a Party's failure to provide a reasonable means of accomplishing delivery. Notice by facsimile or email is effective as of transmission to each and all of the telefacsimile numbers or email addresses provided below for a Party, but, unless receipt is acknowledged by the recipient, must be followed up by notice by registered mail or overnight carrier to be effective. Notice by overnight mail shall be deemed to have been given the Business Day after it is sent, if sent for next day delivery to a domestic address by a recognized overnight delivery service (e.g., Federal Express or UPS). Notice by certified or registered mail, return receipt requested, shall be deemed to have been given upon receipt.



Rocky Mountain Power Exhibit RMP___(MPT-2) Page 49 of 82 Docket No. 18-035-08 Witness: Mark P. Tourangeau





23.2 <u>Changes of Address</u>. The Parties may change any of the persons to whom notices are addressed, or their addresses, by providing written notice in accordance with this section.

SECTION 24 CONFIDENTIALITY

24.1 <u>Confidential Business Information</u>. The following constitutes "Confidential Business Information," whether oral or written: (a) the Parties' proposals and negotiations concerning this Agreement, made or conducted prior to the Effective Date, (b) the terms hereof, (c) information provided under Section 6.9.1, (d) the actual charges billed to PacifiCorp hereunder, and (e) any information delivered by PacifiCorp to Seller prior to the Effective Date relating to the market prices of energy and methodologies for their determination or estimation, and (f) information provided by one Party to the other pursuant hereto. Seller and PacifiCorp each agree to hold such Confidential Business Information wholly confidential. Such Confidential Business Information may only be used by the Parties for purposes related to the approval, administration or enforcement hereof and for no other purpose. "Confidential Business Information that (x) is in or enters the public domain through no fault of the Party receiving such information, and (y) was in the possession of a Party prior to the Effective Date, and (e) above.

A Party providing any Confidential Business Information under this Agreement shall clearly mark all pages of all documents and materials to be treated as Confidential Business information with the term "Confidential" on the front of each page, document or material. If the Confidential Business Information is transmitted by electronic means the title or subject line shall indicate the information is Confidential Business Information. All Confidential Business Information shall be maintained as confidential, pursuant to the terms of Section 24, for a period of two years from the date it is received by the receiving Party unless otherwise agreed to in writing by the Parties.

24.2 Duty to Maintain Confidentiality. Each Party agrees not to disclose Confidential Business Information to any other person (other than its Affiliates, accountants, auditors, counsel, consultants, lenders, prospective lenders, purchasers, insurers, prospective purchasers, contractors providing services to the Facility, employees, officers and directors who agree to be bound by the provisions of this section), without the prior written consent of the other Party, provided that: (a) either Party may disclose Confidential Business Information, if and to the extent such disclosure is required (i) by Requirements of Law, (ii) in order for PacifiCorp to receive regulatory recovery of expenses related to this Agreement, (iii) pursuant to an order of a court or regulatory agency, or (iv) in order to enforce this Agreement or to seek approval hereof, and (b) notwithstanding any other provision hereof, PacifiCorp may in its sole discretion disclose or otherwise use for any purpose in its sole discretion the Confidential Business Information described in Sections 24.1(d) or 24.1(e). In the event a Party is required by Requirements of Law to disclose Confidential Business Information, such Party shall to the extent possible notify the other Party at least three (3) Business Days in advance of such disclosure.

PacifiCorp Regulatory Compliance. The Parties acknowledge that PacifiCorp is 24.3required by law or regulation to report certain information that could embody Confidential Business Information from time to time. Such reports include models, filings, reports of PacifiCorp's net power costs, general rate case filings, power cost adjustment mechanisms, FERC-required reporting such as those made on FERC Form 1, Form 12, or Form 714, market power and market monitoring reports, annual state reports that include resources and loads, integrated resource planning reports, reports to entities such as NERC, WECC, Pacific Northwest Utility Coordinating Committee, WREGIS, or similar or successor organizations, forms, filings, or reports, the specific names of which may vary by jurisdiction, along with supporting documentation. Additionally, in regulatory proceedings in all state and federal jurisdictions in which it does business, PacifiCorp will from time to time be required to produce Confidential Business Information. PacifiCorp may use its business judgment in its compliance with all of the foregoing and the appropriate level of confidentiality it seeks for such disclosures. PacifiCorp may submit Confidential Business Information in regulatory proceedings without notice to Seller if PacifiCorp has obtained in such proceedings a protective order covering such Confidential Business Information.

24.4 <u>Irreparable Injury; Remedies</u>. Each Party agrees that violation of the terms of this Section 24 constitutes irreparable harm to the other, and that the harmed Party may seek any and all remedies available to it at law or in equity, including injunctive relief.

24.5 <u>News Releases and Publicity</u>. Except as otherwise provided in Section 6.13, before either Party issues any news release or publicly distributed promotional material regarding the Facility that mentions the Facility, such Party shall first provide a copy thereof to the other

Party for its review and approval. Any use of either Party's name in such news release or promotional material must adhere to such Party's publicity guidelines then in effect; any use of Berkshire Hathaway's name requires PacifiCorp's prior written consent.

SECTION 25 DISAGREEMENTS

25.1 <u>Negotiations</u>. The Parties shall attempt in good faith to resolve all disputes arising out of, related to or in connection with this Agreement promptly by negotiation, as follows. Any Party may give the other Party written notice of any dispute not resolved in the normal course of business. Executives of both Parties at levels one level above the personnel who have previously been involved in the dispute shall meet at a mutually acceptable time and place within 10 days after delivery of such notice, and thereafter as often as they reasonably deem necessary, to exchange relevant information and to attempt to resolve the dispute. If the matter has not been resolved within 30 days after the referral of the dispute to such senior executives, or if no meeting of such senior executives has taken place within 15 days after such referral, either Party may initiate litigation as provided hereinafter if neither Party has requested that the dispute be mediated in accordance with Section 25.2 below. All negotiations pursuant to this clause are confidential.

25.2 <u>Mediation; Technical Expert.</u>

25.2.1 <u>Mediation</u>. If the dispute is not resolved within 30 days after the referral of the dispute to senior executives, or if no meeting of senior executives has taken place within 15 days after such referral, either Party may request that the matter be submitted to nonbinding mediation. If the other Party agrees, the mediation will be conducted in accordance with the Construction Industry Arbitration Rules and Mediation Procedures (Including Procedures for Large, Complex Construction Disputes) of the American Arbitration Association ("AAA"), as amended and effective on the date of the dispute (the "Mediation Procedures"), notwithstanding any Dollar amounts or Dollar limitations contained therein.

(a) The Party requesting the mediation, may commence the mediation process with AAA by notifying AAA and the other Party in writing ("Mediation Notice") of such Party's desire that the dispute be resolved through mediation, including therewith a copy of the Dispute Notice and the response thereto, if any, and a copy of the other Party's written agreement to such mediation.

(b) The mediation shall be conducted through, by and at the office of AAA located in Salt Lake City, Utah.

(c) The mediation shall be conducted by a single mediator. The Parties may select any mutually acceptable mediator. If the Parties cannot agree on a mediator within five (5) days after the date of the Mediation Notice, then the AAA's Arbitration Administrator shall send a list and resumes of three available mediators to the Parties, each of whom shall strike one name, and the remaining person shall be appointed as the mediator. If more than one name remains, either because one or both Parties have failed to respond to the AAA's Arbitration Administrator within five (5) days after receiving the list or because one or

both Parties have failed to strike a name from the list or because both Parties strike the same name, the AAA's Arbitration Administrator will choose the mediator from the remaining names. If the designated mediator shall die, become incapable or, unwilling to, or unable to serve or proceed with the mediation, a substitute mediator shall be appointed in accordance with the selection procedure described above in this Section 25.2.1, and such substitute mediator shall have all such powers as if he or she has been originally appointed herein.

(d) The mediation shall consist of one or more informal, nonbinding meetings between the Parties and the mediator, jointly and in separate caucuses, out of which the mediator will seek to guide the Parties to a resolution of the Dispute. The mediation process shall continue until the resolution of the dispute, or the termination of the mediation process pursuant to Section 25.2.1(f). The costs of the mediation, including fees and expenses, shall be borne equally by the Parties.

(e) All verbal and written communications between the Parties and issued or prepared in connection with this Section 25.2 shall be deemed prepared and communicated in furtherance, and in the context, of dispute settlement, and shall be exempt from discovery and production, and shall not be admissible in evidence (whether as admission or otherwise) in any litigation or other proceedings for the resolution of the dispute.

(f) The initial mediation meeting between the Parties and the mediator shall be held within 20 days after the Mediation Notice. Either Party may terminate the mediation process upon or after the earlier to occur of (i) the failure of the initial mediation meeting to occur within 20 days after the date of the Mediation Notice, (ii) the passage of 30 days after the date of the Mediation Notice without the dispute having been resolved, or (iii) such time as the mediator makes a finding that there is no possibility of resolution through mediation.

(g) All deadlines specified in this Section 25.2.1 may be extended by mutual agreement.

25.2.2 <u>Technical Expert</u>. If the dispute regards the disputed amount of any invoice, the Parties may, in lieu of mediation, have such dispute resolved pursuant to this Section 25.2.2. Any such dispute will be determined by an independent technical expert, who shall be a mutually acceptable third party with training and experience in the disciplines relevant to the matters with respect to which such person is called upon to provide a certification, evaluation or opinion (the "Technical Expert"), which determination shall be (x) made in accordance with the Construction Industry Arbitration Rules and Mediation Procedures (Including Procedures for Large, Complex Construction Disputes) of the AAA, as amended and effective on the date of the dispute (the "Technical Dispute Procedures"), notwithstanding any Dollar amounts or Dollar limitations contained therein, and (y) binding upon the Parties.

(a) Either Party may commence the technical dispute process with AAA by notifying AAA and the other Party in writing ("Technical Dispute Notice") of such Party's desire that the dispute be resolved through a determination by a technical expert.

(b) The determination shall be conducted by a sole Technical Expert. The Parties may select any mutually acceptable Technical Expert. If the Parties cannot agree on a Technical Expert within five (5) days after the date of the Technical Dispute Notice, then the AAA's Arbitration Administrator shall send a list and resumes of three available technical experts meeting the qualifications set forth in Section 25.2.2 to the Parties, each of whom shall strike one name, and the remaining person shall be appointed as the Technical Expert. If more than one name remains, either because one or both Parties have failed to respond to the AAA's Arbitration Administrator within five (5) days after receiving the list or because one or both Parties have failed to strike a name from the list or because both Parties strike the same name, the AAA's Arbitration Administrator will choose the Technical Expert from the remaining names. If the designated Technical Expert shall die, become incapable or, unwilling to, or unable to serve or proceed with the determination, a substitute technical expert shall be appointed in accordance with the selection procedure described above, and such substitute Technical Expert shall have all such powers as if he or she has been originally appointed herein.

(c) Within 30 days of the appointment of the Technical Expert pursuant to the foregoing sub-section, each Party shall submit to the Technical Expert a written report containing its position with respect to the dispute, and arguments therefor together with supporting documentation and calculations. Discovery shall be limited to Facility documentation relating to the disputed matter. Within 60 days from receipt of such submissions, the Technical Expert shall select one or the other Party's position with respect to the dispute, whereupon such selection shall be a binding determination upon the Parties for all purposes hereof. The costs of the determination by the Technical Expert of any dispute, including fees and expenses, shall be borne by the Party whose position was not selected by the Technical Expert. If the Technical Expert fails to render a decision within 90 days from receipt of each Party's submissions, either Party may initiate litigation in accordance with the provisions herein.

(d) All verbal and written communications between the Parties and issued or prepared in connection with this Section 25.2.2 shall be deemed prepared and communicated in furtherance, and in the context, of dispute settlement, and shall be exempt from discovery and production, and shall not be admissible in evidence (whether as admission or otherwise) in any litigation or other proceedings for the resolution of the dispute.

(e) All deadlines specified in this Section 25.2.2 may be extended by mutual agreement of the Parties.

25.3 <u>Place of Contract Formation; Choice of Forum</u>. Seller and PacifiCorp acknowledge and agree that this Agreement has been made and entered into as of the date first set forth above in the City of Salt Lake City, Utah. Each Party irrevocably consents and agrees that any legal action or proceeding arising out of this Agreement or the actions of the Parties leading up to the Agreement shall be brought exclusively in the United States District Court for the District of Utah in Salt Lake City, Utah, or if such court does not have jurisdiction, in the 3rd Judicial District (Salt Lake County) Court of the State of Utah. By execution and delivery hereof, each Party (a) accepts the exclusive jurisdiction of such court and waives any objection that it may now or hereafter have to the exercise of personal jurisdiction by such court over each Party for the purpose of any proceeding related to this Agreement, (b) irrevocably agrees to be bound by any final judgment (after any and all appeals) of any such court arising out of such documents or actions, (c) irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of venue of any suit, action or proceedings arising out of such documents brought in such court (including any claim that any such suit, action or proceeding has been brought in an inconvenient forum) in connection herewith, and, (d) agrees that service of process in any such action may be effected by mailing a copy thereof by registered or certified mail, postage prepaid, to such Party at its address as set forth herein, and (e) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law.

25.4 <u>Settlement Discussions</u>. No statements of position or offers of settlement made in the course of the dispute process described in this section will be offered into evidence for any purpose in any litigation between the Parties, nor will any such statements or offers of settlement be used in any manner against either Party in any such litigation. Further, no such statements or offers of settlement shall constitute an admission or waiver of rights by either Party in connection with any such litigation. At the request of either Party, any such statements and offers of settlement, and all copies thereof, shall be promptly returned to the Party providing the same.

25.5 <u>Waiver of Jury Trial</u>. EACH PARTY KNOWINGLY, VOLUNTARILY, INTENTIONALLY AND IRREVOCABLY WAIVES THE RIGHT TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION BASED ON THIS AGREEMENT, OR ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT AND ANY AGREEMENT EXECUTED OR CONTEMPLATED TO BE EXECUTED IN CONJUNCTION WITH THIS AGREEMENT, OR ANY COURSE OF CONDUCT, COURSE OF DEALING, STATEMENTS (WHETHER VERBAL OR WRITTEN) OR ACTIONS OF ANY PARTY HERETO. THIS PROVISION IS A MATERIAL INDUCEMENT TO EACH OF THE PARTIES FOR ENTERING HEREINTO. EACH PARTY HEREBY WAIVES ANY RIGHT TO CONSOLIDATE ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER AGREEMENT EXECUTED OR CONTEMPLATED TO BE EXECUTED IN CONJUNCTION WITH THIS AGREEMENT, OR ANY MATTER ARISING HEREUNDER OR THEREUNDER, WITH ANY PROCEEDING IN WHICH A JURY TRIAL HAS NOT OR CANNOT BE WAIVED.

SECTION 26 COMMISSION APPROVAL

A condition precedent to the effectiveness of this Agreement is approval without material modification by the Commission of this Agreement and the REC between University and PacifiCorp. Each Party agrees that it will not petition the Commission for or otherwise seek an order of the Commission to cancel, terminate or modify any provisions of this Agreement or the REC without the prior written consent of the other Party. If a Commission order regarding this Agreement or the REC includes any condition or requirement that is materially adverse to either Party, the Party adversely impacted may terminate this Agreement by providing the other Party written notice within 30 days of the entry of the Commission order, in which case the Parties may but are not required to negotiate revisions or an amendment to this Agreement with mutually acceptable rates, terms and conditions, which amendment or revised agreement shall be subject to Commission approval.

SECTION 27 MATERIAL TERMS AND CONDITIONS – UNIVERSITY

PacifiCorp and Seller represent that, before entering into this Agreement, they secured University's consent to material terms and conditions included in this Agreement related to resource size, estimated production, expected Delivery Start Date, contract Term, the costs and charges for Output, performance guarantees, operational terms, credit provisions, and default and termination provisions. University is a third-party beneficiary of this Agreement and has the right, but not the obligation, to enforce the Agreement according to its terms. University may, in its sole discretion, cure any default by either Party under this Agreement and recover from the defaulting Party any damages, costs or expenses suffered or incurred by University in enforcing this Agreement or curing a default, including reasonable attorneys' fees. PacifiCorp and Seller each agrees to provide written notice to University of any default by the other Party under this Agreement that could lead to termination as soon as practicable and in all events prior to the expiration of the cure period applicable to such default.

[Signature page(s) to follow]

Rocky Mountain Power Exhibit RMP___(MPT-2) Page 57 of 82 Docket No. 18-035-08 Witness: Mark P. Tourangeau

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed in their respective names as of the date first above written.

AMOR IX, LLC

PACIFICORP

By: michalas Goodman	By:
Name: MICHOLAS GOOMAN	Name:
Title: CEO	Title:

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed in their respective names as of the date first above written.

AMOR IX, LLC	PACIFICORP
By:	ву: М-/
Name:	Name: Mark Tourangeau
Title:	Title: Director Commercial Services

Rocky Mountain Power Exhibit RMP___(MPT-2) Page 59 of 82 Docket No. 18-035-08 Witness: Mark P. Tourangeau

EXHIBIT A

EXPECTED ENERGY

FIRST 12 MONTHS FOLLOWING DELIVERY START DATE, TO BE UPDATED ANNUALLY



Amor will update Exhibit A annually by November 30 of each year for review by University. In no instance will the monthly or annual totals in Exhibit A change by more than plus or minus 3% from the previous year without the written agreement of Amor and University.

EXHIBIT B

Soda Lake Facility and Premises AMOR IX, LLC

DESCRIPTION OF FACILITY AND PREMISES

Generating Facility Description: The generating Facility consists of one (1) Ormat Energy Convertor (OEC) with a nameplate capacity of 20,000 kW Net Output. The OEC converts heat provided by geothermal fluids into electricity utilizing the Organic Rankin Cycle. The OEC equipment consist of evaporators, condensers, turbines, generators, cycle-pumps, system controls, control valves, and piping.

The generating Facility is supplied with geothermal fluids from a well field developed by Seller. The geothermal fluids are produced and transported from the geothermal production wells to the power plant. The fluids flow under pressure directly into the OEC evaporators. The geothermal fluids heat and vaporizes the working fluid, which expands through the turbines, which are coupled to the generators. The expanding working fluids are condensed by air cooled fans and pumped back into to the heat exchangers in a closed loop. The spent geothermal fluids are returned via pipelines from the generating Facility to the injection wells.

Surface Facilities: Soda Lake is located 4 miles northwest of Fallon, Nevada. The geothermal Facility is located within Section 33, T20N, R28E, M.D.M., Churchill County, Nevada. The property owner is Truckee-Carson Irrigation District and the surface areas are leased.

Wellfield: Soda Lake's active and standby wells in the geothermal wellfield are situated within the following Sections or portions of Sections. Geothermal resources are leased. The geothermal resource leases—both active leases and buffer leases—are held within the Soda Lake Geothermal Unit.

Geothermal Well Locations: N¹/₂ Sec 28, T20N, R28E (BLM Lease N-11737); S¹/₂ Sec 28, T20N, R28E (BLM Lease N-53371); Sec 29, T20N, R28E (Powelson Lease); and Sec 33, T20N, R28E (TCID Lease).

Transmission: Soda Lake's Transmission Line is situated on 13 different easement parcels. The easement grantors and the easement descriptions are listed below:

Parcel No.	Easement Grantor	Notes
(Per Title		
Policy)		
Parcel 1	Truckee-Carson Irrigation	Transmission Line Right of Way Grant
	District	(11/09/1987)
Section 33, T20N, R28E, M.D.M. Commencing at the E ¹ / ₄ corner of said Section 33, and		
proceeding thence along the East line of the NE ¹ / ₄ of said Section 33, N 00°40'39" W 12.50'.		
Thence S 89°19'21" W 250.00' to the true point of beginning, the said point lying on the		
Westerly line of the proposed geothermal plant site. Thence proceeding along the centerline of a		

25 foot wide right-of-way said centerline lying 12.5' North of the East-West center of section line of said Section 33 S 89°19'21" W 5047.71' to the point of ending, said point lying on the West line of said Section 33. The sidelines of said right-of-way shall be prolonged or shortened so as to terminate on the end lines. Together with 5 foot by 45 foot anchor easements as required at each angle point.

Parcel 2	Truckee-Carson Irrigation	Transmission Line Right of Way Grant
	District	(11/09/1987)

NW ¼ of Section 32, T20N, R28E, M.D.M. Beginning at a point on the East line of the NW ¼ of said Section 32, said point being 12.5' North of the SE corner of said NW ¼ and proceeding thence along the centerline of a 25 foot wide right-of-way S 89°34'31" W 2646.63' to the point of ending, said point lying on the West line of said NW ¼, from which point the West ¼ corner of said Section 32 bears S 12.5'. The sidelines of said right-of-way shall be prolonged or shortened so as to terminate on the end lines. Together with 5 foot by 45 foot anchor easements as required at each angle point.

Parcel 3	Truckee-Carson Irrigation	Transmission Line Right of Way Grant
	District	(11/09/1987)

The N $\frac{1}{2}$ SW $\frac{1}{4}$ Section 2, T19N, R27E, M.D.M. Beginning at a point on the North line of the N $\frac{1}{2}$ of the SW $\frac{1}{4}$ of said Section 2 from which point the W $\frac{1}{4}$ corner of said Section 2 bears West 12.5'. Thence proceeding along the centerline of a 25 foot wide right-of-way S 00°27'42'' W 1321.68' to the point of ending, said point lying on the South line of the N $\frac{1}{2}$ of the SW $\frac{1}{4}$ of said Section 2. Together with 5 foot by 45 foot anchor easements as required at each angle point.

Parcel 4	Truckee-Carson Irrigation	Transmission Line Right of Way Grant
	District	(11/09/1987)

Section 11, T19N, R27E, M.D.M. Beginning at a point on the North line of the NW ¼ of said Section 11 said point lying 12.5' East of the NW corner of said Section 11 and proceeding thence along the centerline of a 25 foot right-of-way S 05°19'22" W 135.79' to the point of ending, said point lying on the West line of the NW ¼ of said Section 11. The sidelines of said right-of-way shall be prolonged or shortened so as to terminate on the end lines. Together with 5 foot by 45 foot anchor easements as required at each angle point.

Parcel 5	Truckee-Carson Irrigation	Transmission Line Right of Way Grant
	District	(11/09/1987)

Section 10, T19N, R27E, M.D.M. Beginning at a point on the North line of the S $\frac{1}{2}$ of the SE $\frac{1}{4}$ of said Section 10, thence along the centerline of a 25 foot right-of-way S 00°06'14" W 1323.54' to the point of ending, said point lying on the South line of the SE $\frac{1}{4}$ of said Section 10 from which point the SE corner of said Section 10 bears East 12.5'. The sidelines of said right-of-way shall be prolonged or shortened so as to terminate on the end lines. Together with 5 foot by 45 foot anchor easements as required at each angle point.

Parcel 6	Truckee-Carson Irrigation	Transmission Line Right of Way Grant
	District	(11/09/1987)

Section 15, T19N, R27E, M.D.M. Beginning at a point on the North line of the NE ¼ of said Section 15, from which point the NE corner of said Section 15 bears East 12.5'. Thence along the centerline of a 25 foot right of way S 00°06'42" E 822.14'. Thence S 02°21'35" W 1,102.10'. Thence S 00°06'47" E 180.00'. Thence S 57°39'38" W 588.51' to a point on the Northeasterly line of a Southern Pacific Transportation Company right-of-way the point of ending. The sidelines of said right-of-way shall be prolonged or shortened so as to terminate on the end lines. Together with a 5 foot by 45 foot anchor easements as required at each angle point.

Parcel 7	Margaret E. Oser Trustee	Right of Way Grant (09/11/1987)
		¹ / ₄ of Section 31, T20N, R28E, M.D.M., from
which point the E ¹ / ₄ corner of said Section 31 bears South 12.5'. Thence proceeding along said		
centerline S 89°10'40" W 5,133.60', more or less, to a point on the West line of the NW ¼ of		
		h point the W $\frac{1}{4}$ corner of said Section 31 bears
		strip shall be prolonged or shortened so as to
terminate on t	č	surp shan be protonged of shortened to us to
Parcel 8	Margaret E. Oser Trustee	Right of Way Grant (09/11/1987)
		W ¹ / ₄ of the NW ¹ / ₄ of Section 1, T19N, R27E,
	-	id Section 1 bears S 89°48'39" W 140.70';
		4'39" W, 141.37', to a point on the West line of
		point of ending, from which point the NW corner
	· · · ·	es of said right-of-way strip shall be prolonged or
	as to terminate on the end lines.	es of sala right of way ship shall be protonged of
Parcel 9	U.S. Bureau of Reclamation	Right-of-way Agreement for Use of
		Reclamation Land— Contract No. 0-07-20-
		L5296 (06/21/1990)
Section 32. T	20N, R28E, M.D.M. Commencing	g at the E ¹ / ₄ corner of Section 32, T20N, R28E,
		line of the NE ¹ / ₄ of said Section 32, North 12.5'
· · ·		on the centerline of a 25 foot wide transmission
1		d centerline S 89°34'31" W 2,662.71', more or
-		said Section 32, the point of ending; The sideline
-		tened so as to terminate on the end lines.
-		nts as required at each angle point.
Parcel 10	U.S. Bureau of Reclamation	Right-of-way Agreement for Use of
		Reclamation Land— Contract No. 0-07-20-
		L5296 (06/21/1990)
Section 36, T	20N, R27E, M.D.M. Commencing	g at the E ¹ / ₄ corner of Section 36, T20N, R27E,
M.D.M., and	proceeding thence North 12.5' to	the true point of beginning; Thence proceeding
		sion line right-of-way S 89°10'40" W, 13.02';
Thence S 00°12'56" E, 2,531.33'; thence S 44°09'38" W, 190.89'; thence S 89°48'39" W,		
4,852.36'; thence S 88°44'39" W, 140.99' to a point on the South line of the SW ¼ of said		
Section 36, the point of ending, from which point, the SW corner of said Section 36 bears S		
89°48'39" W, 140.78'; The sidelines of said right-of-way shall be lengthened or shortened so as		
to terminate on the end lines. Together with 5 foot by 45 foot anchor easements as required at		
each angle po		T
Parcel 11	U.S. Bureau of Reclamation	Right-of-way Agreement for Use of
		Reclamation Land— Contract No. 0-07-20-
		L5296 (06/21/1990)
Section 2, T19N, R27E, M.D.M. Commencing at the Northeast corner of Section 2, T19N,		
R27E, M.D.M., and proceeding thence South 12.5' to the true point of beginning, said point		
lying on the centerline of a 25 foot wide transmission line right-of-way; Thence proceeding		
along said centerline N 88°57'15" W, 2,621.43'; Thence continuing along said centerline N		
89°07'45" W, 2,649.33'; Thence S 00°40'15" W, 2,684.47', to a point on the South line of the		
NW ¼ of said Section 2, the point of ending; The sidelines of said right-of-way shall be		
lengthened or shortened so as to terminate on the end lines. Together with 5 foot by 45 foot		
anchor easem	ents as required at each angle poin	nt.
-----------------	-------------------------------------	--
Parcel 12	U.S. Bureau of Reclamation	Right-of-way Agreement for Use of
		Reclamation Land— Contract No. 0-07-20-
		L5296 (06/21/1990)
Section 2, T1	9N, R27E, M.D.M. Commencing	at the NW corner of the S ¹ / ₂ of the SW ¹ / ₄ of
Section 2, T1	9N, R27E, M.D.M., and proceeding	ng thence E 12.5' to the true point of beginning,
said point lyin	ng on the centerline of a 25 foot w	vide transmission line right-of-way; Thence
1 0	0	W, 1,321.14' to a point the South line of the SW $\frac{1}{4}$
of said Section	on 2, the point of ending, from whi	ch point the SW corner of said Section 2 bears W
	.	be lengthened or shortened so as to terminate on
the end lines.	Together with 5 foot by 45 foot a	nchor easements as required at each angle point.
Parcel 13	U.S. Bureau of Reclamation	Right-of-way Agreement for Use of
		Reclamation Land— Contract No. 0-07-20-
		L5296 (06/21/1990)
Section 10, T	19N, R27E, M.D.M. Commencin	g at the NE corner of Section 10, T19N, R27E,
	1 0	point on the centerline of a 25 foot wide
transmission	line right-of-way, the point of beg	inning; Thence along said centerline S 05°19'22"
W, 135.40'; t	hence S 00°01'36" W, 2,373.09';	thence S 00°06'14" W, 1,322.27' to a point on
		ection 10, the point of ending; The sidelines of
0		ned so as to terminate on the end lines. Together
with 5 foot by	y 45 foot anchor easements as requ	uired at each angle point.

EXHIBIT B — Attachments

1. Site map



EXHIBIT C

NERC EVENT TYPES

Event	Description of Outages
Туре	
U1	<u>Unplanned (Forced) Outage—Immediate</u> – An outage that requires immediate removal of a unit from service, another outage state or a Reserve Shutdown state. This type of outage results from immediate mechanical/electrical/hydraulic control systems trips and operator-initiated trips in response to unit alarms.
U2	<u>Unplanned (Forced) Outage—Delayed</u> – An outage that does not require immediate removal of a unit from the in-service state but requires removal within six (6) hours. This type of outage can only occur while the unit is in service.
U3	<u>Unplanned (Forced) Outage—Postponed</u> – An outage that can be postponed beyond six hours but requires that a unit be removed from the in-service state before the end of the next weekend. This type of outage can only occur while the unit is in service.
SF	<u>Startup Failure</u> – An outage that results from the inability to synchronize a unit within a specified startup time period following an outage or Reserve Shutdown. A startup period begins with the command to start and ends when the unit is synchronized. An SF begins when the problem preventing the unit from synchronizing occurs. The SF ends when the unit is synchronized or another SF occurs.
МО	<u>Maintenance Outage</u> – An outage that can be deferred beyond the end of the next weekend, but requires that the unit be removed from service before the next planned outage. (Characteristically, a MO can occur any time during the year, has a flexible start date, may or may not have a predetermined duration and is usually much shorter than a PO.)
ME	<u>Maintenance Outage Extension</u> – An extension of a maintenance outage (MO) beyond its estimated completion date. This is typically used where the original scope of work requires more time to complete than originally scheduled. Do not use this where unexpected problems or delays render the unit out of service beyond the estimated end date of the MO.
РО	<u>Planned Outage</u> – An outage that is scheduled well in advance and is of a predetermined duration, lasts for several weeks and occurs only once or twice a year. (Boiler overhauls, turbine overhauls or inspections are typical planned outages.)
PE	<u>Planned Outage Extension</u> – An extension of a planned outage (PO) beyond its estimated completion date. This is typically used where the original scope of work requires more time to complete than originally scheduled. Do not use this where unexpected problems or delays render the unit out of service beyond the estimated end date of the PO.

Rocky Mountain Power Exhibit RMP___(MPT-2) Page 66 of 82 Docket No. 18-035-08 Witness: Mark P. Tourangeau

EXHIBIT D

CONTRACT PRICE (\$/MWh)

			-	
		_		
 		_		
		_		

(T)	
E	
A	
H	
E	

<u>Soda Lake Permits</u> <u>AMOR IX, LLC</u>

rermit holder	Type of Permit	Permit Number	Purpose	
AMOR IX, LLC	NDEP Underground Injection Control Permit (UIC)	UNEV89037 (05/08/2014)	Operation of injection wells	
AMOR IX, LLC	NV Division of Minerals Geothermal Resource Development Permit	1232 (05/12/2011)	Observation well 44B-34 (not currently in use)	
AMOR IX, LLC	NV Division of Minerals- Geothermal Resource Development Permit	1142 (06/03/2010)	Production well 25A-33	
AMOR IX, LLC	NV Division of Minerals- Geothermal Resource Development Permit	913 (05/15/2009)	Production well 41B-33 (not currently in use)	
AMOR IX, LLC	NV Division of Minerals- Geothermal Resource Development Permit	912 (05/15/2009)	Production well 45A-33 (not currently in use)	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	369 (09/03/1993)	Production well 32-33	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	269 (05/29/1991)	Production well 41-33	Witness:
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	271 (06/05/1991)	Production well 41A-33	Mark P. To

Permit Holder	Type of Permit	Permit Number	Purpose	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	265 (02/08/199)	Injection well 87-29	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	266 (02/21/1991)	Injection well 45-28 (currently on standby)	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	263 (01/09/1991)	Observation well 53-28 (not currently in use)	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	268 (04/22/1991)	Injection well 81-33	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	239 (02/13/1990)	Observation well 64-33 (not currently in use)	
AMOR IX, LLC	NV Department of Minerals Geothermal Resource Development Permit	247 (07/17/1990)	For well 84-33A (currently on standby)	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	250 (09/19/1990)	Production well 84B-33	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	259 (11/29/1990)	Observation well 55-33 (not currently in use)	
AMOR IX, LLC	NV Department of Minerals- Geothermal Resource Development Permit	1411 (11/04/2016)	Observation well 58-34 (not currently in use). This well pre-dated NDOM and was initially permitted by DWR	
				-

	r	r			r	Witness: I
				12/19/2014 Email from Lorenzo Trimble verifies that GDPs remain in original permittee's name		12/19/2014 Email from Lorenzo Trimble verifies that GDPs remain in original permittee's name
Purpose	Production well 84-33 (not currently in use). This well pre-dated NDOM and was initially permitted by DWR	Injection well 77-29. This well pre-dated NDOM and was initially permitted by DWR	Statewide Bond	Observation well 44B-34 (not currently in use)	Injection well 45-28 (currently on standby)	Observation well 53-28 (not currently in use)
Permit Number	DWR41931 (04/16/1981)	404 (06/02/1994)	Bond No. RLB0015954 (02/10/2015)	(05/21/2011)	(02/19/1991)	(08/16/1990)
Type of Permit	NV Department of Minerals- Geothermal Resource Development Permit	NV Department of Minerals- Geothermal Resource Development Permit	NV Department of Minerals- Geothermal Statewide Drilling Surety	BLM Geothermal Drilling Permit	BLM Geothermal Drilling Permit	BLM Geothermal Drilling Permit
Permit Holder	AMOR IX, LLC f/k/a AMOR IX Corporation	AMOR IX, LLC	Raser Power Systems and AMOR IX, LLC	SLRP	AMOR IX. LLC	Ormat Energy Systems Inc.

Permit Holder	Type of Permit	Permit Number	Purpose	
Western States Geothermal Co.	BLM Geothermal Drilling Permit	(05/19/1998)	Observation well 58-34 (not currently in use)	12/19/2014 Email from Lorenzo Trimble verifies that GDPs remain in original permittee's name.
Magma Energy	BLM Geothermal Drilling Permit	(07/30/2009)	Production well 41B(88-32)- 33 (directional drilling into federal lands). Not currently in use.	12/19/2014 Email from Lorenzo Trimble verifies that GDPs remain in original permittee's name.
Raser Power Systems and AMOR IX, LLC	BLM Geothermal Statewide Drilling Surety Bond	Bond No. RLB0015955 (03/24/2015)	Statewide Bond	
AMOR IX, LLC	Soda Lake Unit Agreement	N-13204-X (01/01/1996)	Unit Agreement for Soda Lake	
AMOR IX, LLC	Churchill County Special Use Permit	(10/12/2016)	Allows 24 MW Binary Geothermal Power Plant (Repower/ SL-3)	
AMOR IX, LLC	Churchill County Business License	No. 11673 (01/01/2016)		
AMOR IX, LLC	NV Division of Water Resources Certificate of Appropriation of Water	Permit No. 60927 (Certificate No. 15897)	SL1 cooling tower make up water well.	<u>Vitness: Mark P</u>
AMOR IX, LLC	NV Division of Water Resources Permit for Appropriation of Water	Permit No. 60928	SL1 cooling tower make up water well.	_ Lourangea
				L

Permit Holder	Type of Permit	Permit Number	Purpose	
AMOR IX, LLC	NV Division of Water Resources Certificates of Appropriation of Water	Permit# (Certificate#) 41931 (13576) 59592 (14803) 59593 (14804)	For geothermal production wells	
		59594 (14805) 59595 (14806)		
AMOR IX, LLC	NV State Fire Marshall Hazardous Materials Permits	Company ID# 2272	To permit and control safe handling of hazardous	
		Facility ID #s 4796 and 4797	chemicals on site (SL-1 and SL-2)	
AMOR IX, LLC	NV Dept. of Business and	Chain of	Operating permits for all	
	Industry - Boiler and Pressure Vessel Operating Permit	permits to AMOR IX,	pressure vessels on site.	
)	LLC dated		
		and 00/00/20		
AMOR IX Corporation	NDEP Class II Air Quality	AP4911-	To control fugitive emissions,	This permit has been
n/k/a AMOR IX, LLC	Operating Permit	0464.04	dust, pentane and diesel	amended and
	(Superseded by Amendment Relow)	(12/19/2014)	storage tanks and exhaust. Includes Surface Area	reissued to allow SI -3 Renower
			Disturbance Permits.	

Rocky Mountain Power Exhibit RMP___(MPT-2) Page 71 of 82 Docket No. 18-035-08 Witness: Mark P. Tourangeau

		1	1	r	· · · · ·	Nitness: Mark P.
Purpose	Required for construction of SL-3; Surface Area Disturbance (SAD) approval has been added to AMOR's prior Class II Permit	This facility pre-dates the CAPP regulations so it has no permit, but still must register and report to NDEP	Original state construction permit for Soda Lake II	Notice of Self Certification of QF Status	Operate hand held radios	Construction of septic system for O&M Building
Permit Number	Revision of Class II Air Quality Operating Permit AP4911- 0464.04, Air Case 9149 (05/08/2017)	Facility Tracking ID# 42 FIN: A0548	UEPA No. 189 Docket No. 89-959	Notice in Docket QF87- 263-005 (11/09/2016)	Call No. WQBC239 (07/31/2014)	GNEV SODS09 S0080 (12/15/2011)
Type of Permit	NDEP Class II Air Quality Operating Permit Amendment	Chemical Accident Prevention Program (CAPP)	NV Public Service Comm'n UEPA Permit to Construct	FERC	FCC Radio Station Authorization	NDEP-BWPC Construction of On-Site Disposal System (OSDS) Permit
Permit Holder	AMOR IX, LLC	AMOR IX, LLC	AMOR IX Corporation n/k/a AMOR IX, LLC	AMOR IX, LLC	AMOR IX, LLC	Lumos and Associates (construction contractor)

Permit Holder	Type of Permit	Permit Number	Purpose
	Chemical Accident Prevention Program (CAPP)	In process	Required prior to the operation of SL-3
	Building and construction permits and related state/local approvals for SL-3	Not applied for (EPC responsibility)	Building and construction

In addition to the Permits listed at Exhibit E, Permits required for SL-3 are:

EXHIBIT F

REQUIRED FACILITY DOCUMENTS

1. <u>Obtained Required Facility Documents</u>:

Licenses, Permits and Authorizations:

As identified on Exhibit E and as contained in the other documents listed on this Exhibit F.

Land Rights:

2. <u>To Be Obtained Required Facility Documents:</u>

Licenses, Permits and Authorizations:

As identified on Exhibit E and as contained in the other documents on this Exhibit F.

Operation and Maintenance:

EPC Agreement was executed by and between Amor IX, LLC and Ormat Nevada Inc., on November 30, 2016.

Once the Facility achieves commercial operation Amor IX, LLC will take control of the Facility and provide all operation and maintenance.

The Large Generator Interconnection Agreement was executed on March 13, 2017 by and between Amor IX, LLC and Sierra Pacific Power Company d/b/a NV Energy.

The Facility has a backup generator on site and there is no retail service agreement required.

Proof of Insurance: (See Exhibit J)

Operations and Maintenance Agreements:

Provided by Ormat

SUCH LIST MAY BE UPDATED PURSUANT TO SECTION 3.2.4

EXHIBIT G

<u>Soda Lake Site and Producing or Injecting Resource Leases</u> <u>AMOR IX, LLC</u>

Original Lessor	Site Lease Description	Property Description	Notes
Truckee-Carson Irrigation District (TCID)	Title Policy Lease Parcel #1 Originally Recorded #233154 (Lease Date 11/09/1987)	Within Section 33, T20N, R28E, M.D.M., Churchill County, Nevada	Site of SL-1
Truckee-Carson Irrigation District (TCID)	Title Policy Lease Parcel #2 Originally Recorded #251452 (Lease Date 03/01/1990)	Within Section 33, T20N, R28E, M.D.M., Churchill County, Nevada	Site of SL-2 and SL-3 Amended on 10/26/2016 to expand the site to include the SL-3 footprint. Recorded #457498
Truckee-Carson Irrigation District (TCID)	No Lease Parcel # Originally Recorded #255132 (Lease Date 03/01/1990)	Within Section 33, T20N, R28E, M.D.M., Churchill County, Nevada	Site of Office, Control Center, and Workshop

Notes	Geothermal Well(s) Present on	Geothermal Well(s) Present on	Geothermal Well(s) Present on	Geothermal Well(s) Present on
	this Lease	this Lease	this Lease	Lease N-53371
Property Description	Lots 1, 2, and S ½ NE ¼, Section 4, T19N, R28E, and N ¼, Section 28, and all of Section 34, T20N, R28E, M.D.M., Churchill County, Nevada	SW ¹ / ₄ , Section 29, and NW ¹ / ₄ and E ¹ / ₂ SW ¹ / ₄ of Section 32, and All of Section 33, T20N, R28E, M.D.M., Churchill County, Nevada	SE ¹ / ₄ , Section 29, excepting the W ¹ / ₂ of the NW ¹ / ₄ thereof, T20N, R28E, M.D.M., Churchill County, Nevada	S ½, Section 28, and E ½ and W ½ SW ¼, Section 32, T20N, R28E, M.D.M., Churchill County, Nevada
Resource Lease Description	Resource Parcel (Unit Map) #128 Originally Recorded #251633 (Lease Date 10/01/1975)	Resource Parcel (Unit Map) #119 Originally Recorded #159133 (Lease Date 05/01/1978)	Resource Parcel (Unit Map) #120 Originally Recorded #163196 (Lease Date 03/01/1979)	Resource Parcels (Unit Map) #129 and #130 Originally Recorded #251633 (Lease Date 10/01/1975)
Original Lessor	BLM	Truckee-Carson Irrigation	Robert Powelson and	BLM
	(Lease N-11737)	District (TCID)	Suzanne M.L. Henderson	(Leases N-53371 and 53372)

EXHIBIT H

PACIFICORP'S INITIAL DESIGNATED REPRESENTATIVES

Authorized Representatives:

PacifiCorp:	Director, Origination, Energy Supply Management PacifiCorp 825 NE Multnomah St., Suite 600 Portland, OR 97232-2315 Fax 503-813-6271
With a copy to:	Manager, Marketing and Trading Contracts PacifiCorp Energy Supply Management 825 NE Multnomah St., Suite 600 Portland, OR 97232-2315 Fax 503-813-6291 <u>cntadmin@pacificorp.com</u>

EXHIBIT I

POINT OF DELIVERY/INTERCONNECTION FACILITIES

- 1. The point of metering shall be the point where energy and demand meters are located for the purposes of determining the energy generated by the Facility. The point of metering shall occur at the Facility taking into account losses between Facility and Point of Interconnection.
- 2. The Point of Interconnection shall be the Ragtown Substation on the NV Energy transmission system.
- 3. Point of Delivery: Gon.PAV located near Utah/Nevada border.
- 4. Single-line diagram is attached.
- 5. See attached path one-line. Transmission Service is contracted from the Ragtown Substation to Gon.PAV. Transmission Service Agreement is by and between Amor IX, LLC and NV Energy.
- 6. All Station Service on site is served by the Generating Facility.
- 7. Maximum Transmission Rate 20MW.

EXHIBIT I – Attachments



1. Substation Metering One-Line Diagram

2. Transmission Path



Transmission Path

EXHIBIT J

REQUIRED INSURANCE

1.1 <u>Required Policies and Coverages</u>. Without limiting any liabilities or any other obligations of Seller under this Agreement, Seller shall secure and continuously carry with an insurance company or companies rated not lower than "A-/VII" by the A.M. Best Company the insurance coverage specified below:

1.1.1 Workers' Compensation. Seller shall comply with any applicable laws or statutes, state or federal jurisdiction, where Seller performs work.

1.1.2 Employers' Liability. Seller shall maintain employers' liability insurance covering bodily injury with limits of not less than: \$500,000 – for each accident, \$500,000 for disease – each employee, and \$500,000 for disease – policy limit.

1.1.3 Commercial General Liability. Seller shall maintain insurance to include premises and operations, contractual liability, with a single limit of not less than \$1,000,000 each occurrence to protect against and from loss by reason of injury to persons or damage to property based upon and arising out of the activity under this Agreement in accordance with Seller's obligations in Section 12. Indemnification and Liability.

1.1.4 Business Automobile Liability. Seller shall secure and continuously carry business automobile liability insurance with a single limit of not less than \$1,000,000 each accident covering bodily injury and property damage with respect to Seller's vehicles whether owned, hired or non-owned.

1.1.5 Umbrella/excess Liability. Seller shall maintain umbrella or excess liability insurance on an occurrence and following form basis with a with limits not less than (a) or (b) as follows:

- (a) Facility Capacity Rating under 200 KW \$1,000,000
- (b) Facility Capacity Rating at or above 200 KW \$5,000,000

1.1.6 Property Insurance. Seller shall maintain property insurance covering equipment and structures in an amount at least equal to the Maximum Foreseeable Loss value of Seller's Facility for "all risks" of physical loss or damage, including coverage for earth movement, flood, boiler and machinery. The policy may contain separate sub-limits and deductibles subject to insurance company underwriting guidelines. Property insurance will be maintained in accordance with terms available in the insurance market for similar facilities.

1.2 Additional Provisions or Endorsements:

1.2.1 Except for workers' compensation and property insurance, the polices required herein shall include provisions or endorsements as follows:

(a) PacifiCorp, parent, divisions, officers, directors and employees shall be included as additional insureds;

(b) include provisions that such insurance is primary insurance with respect to the interests of PacifiCorp and that any other insurance maintained by PacifiCorp is excess and not contributory insurance with the insurance required hereunder, and

(c) cross liability coverage or severability of interest.

1.2.2 Unless prohibited by applicable law, all required insurance policies shall contain provisions that the insurer will have no right of recovery or subrogation against PacifiCorp.

1.3 <u>Certificates</u>. Prior to connection of the Facility to PacifiCorp's electric system, or another utility's electric system if delivery to PacifiCorp is to be accomplished by wheeling, Seller shall secure and continuously carry insurance in compliance with the requirements of this Section. Seller shall provide PacifiCorp insurance certificate(s) confirming Seller's compliance with the insurance requirements hereunder. Insurance certificate confirming compliance shall be provided to PacifiCorp by Seller at least annually and each time a new insurance policy is issued or becomes effective. In the event Seller receives notice of cancellation of any insurance required herein for any reason, Seller agrees to notify PacifiCorp within five (5) days of receipt of such notice, and will provide proof of reinstatement or replacement of coverage prior to the effective date of cancellation.

1.4 Commercial General Liability coverage written on a "claims-made" basis, if any, shall be specifically identified on the certificate, and Seller shall be maintained by Seller for a minimum period of five (5) years after the completion of this Agreement and for such other length of time necessary to cover liabilities arising out of the activities under this Agreement.

1.5 <u>Periodic Review</u>. PacifiCorp may review this schedule of insurance as often as once every two (2) years. PacifiCorp may in its discretion require Seller to make reasonable changes to the policies and coverages described in this Exhibit to the extent reasonably necessary to cause such policies and coverages to conform to the insurance policies and coverages typically obtained or required for power generation facilities comparable to the Facility at the time PacifiCorp's review takes place.



Signature & Attestation

State of Nevada **Renewable Energy Tax Abatements Application** AFN:

Attestation and Signature

I, ____John Perry____, by signing this Application, I do hereby attest and affirm under penalty of perjury the following:

(1) I have the legal capacity to submit this Application on behalf of the applicant;

(2) I have prepared and personally knowledgeable regarding the contents of this Application; and (3) The content of this Application are true, correct, and complete.

John Perry Name of person authorized for signature:

9/25/2018

Signature:

CFO Title:

Date:



Attachment 15 Confidential Information

State of Nevada Renewable Energy Tax Abatement Application AFN:

This Application contains confidential information: Yes X_ No

If yes, please identify any information in the within Application or documents submitted herewith, which Applicant considers confidential or trade secret information. Further, identify: (1) the applicable statutory authority or agreement preventing public disclosure of the information; and (2) Applicant's rationale underlying non-disclosure of the information or document(s).

Applicant acknowledges that the burden of demonstrating confidentiality or trade secret status lies with the Applicant, and Applicant agrees to defend and indemnify the State and its agencies for honoring such designation. Notwithstanding, Applicant understands that the over-inclusive designation of information or documents as confidential or trade secret may cause the Nevada State Office of Energy to conduct further inquiry of the Applicant into the confidentiality of the information, potentially delaying submission of the Application to the Nevada Energy Director.

Material for which confidentiality is claimed :

As Marked

Basis for claims of confidentiality:

The Information redacted contains trade secrets and other information that is proprietary and its release would harm AMOR IX, LLC and it is, therefore, confidential pursuant to NRS 360.247, 49.325, and 703.190.